RESOLUTION _____

RESOLUTION OF THE MAYOR AND CITY COUNCIL OF THE CITY OF CARTERSVILLE APPROVING AND AUTHORIZING THE EXECUTION OF A POWER PURCHASE CONTRACT BETWEEN THE SOLAR PARTICIPANT AND THE MUNICIPAL ELECTRIC AUTHORITY OF GEORGIA, THE PLEDGE OF THE FULL FAITH AND CREDIT OF THE SOLAR PARTICIPANT TO SECURE ITS PAYMENT OBLIGATIONS THEREUNDER, AND FOR SUCH OTHER PURPOSES.

WHEREAS, pursuant to the Municipal Electric Authority Act (the "Act"), the City of Cartersville (the "Solar Participant") has previously entered into one or more Power Sales Contracts (each, as amended, a "Power Sales Contract") with the Municipal Electric Authority of Georgia (the "Authority") for provision of the Solar Participant's bulk electric power supply needs by the Authority from defined projection projects and sources; and

WHEREAS, under one such Power Sales Contract, the Project One Power Sales Contract (the "**Project One Power Sales Contract**"), the Authority further agreed to provide or cause to be provided additional power needs of the Solar Participant in excess of its entitlement to power supplied under the Project One Power Sales Contract ("**Supplemental Power**"); and

WHEREAS, the Project One Power Sales Contract provides that the Solar Participant may elect to procure an alternate source of Supplemental Power other than that provided by the Authority from the output of an Authority project; and

WHEREAS, the Authority adopted a Supplemental Power Policy (the "**Supplemental Power Policy**") under which the Solar Participant and the Authority may make elections regarding provision and procurement of Supplemental Power; and

WHEREAS, the Solar Participant has determined that, in order to meet the growing and diverse energy needs of its customers, it has need for an additional type of economical, reliable source of electric power and energy beyond that provided from the sources available resources of the Authority under the Project One Power Sales Contract and other contracts between the City and the Authority; and

WHEREAS, the Authority has informed the Solar Participant that the Authority has an opportunity to procure a substantial amount of Supplemental Power for a multi-year term through a Power Purchase Agreement with Pineview Solar LLC (the "Company") for the output and services of approximately 80 MWac from a photovoltaic solar energy generation facility located in Wilcox County, Georgia (the "Facility") to be constructed, owned, operated, and maintained by the Company (such agreement, the "Supplemental Power Purchase Agreement" or "SPPA"); and

WHEREAS, in accordance with the Supplemental Power Policy, the Solar Participant has requested that the Authority purchase from the Company power, output and services of the Facility to cause to be provided to the City its Supplemental Power; and

WHEREAS, the Authority has agreed to cause to be provided the Solar Participant's Supplemental Power from the power, output and services of the Facility pursuant to the terms of a Power Purchase Contract (the "**PPC**") in substantially the form attached as <u>Exhibit A</u> hereto; and

WHEREAS, the Solar Participant finds, and the Solar Participant and the Authority agree that the PPC is supplemental to, and is authorized by, the Project One Power Sales Contract and that the Products (as defined in the SPPA) constitute Supplemental Power as defined in the Supplemental Power Policy; and

WHEREAS, the Solar Participant determines that the Solar Participant's payment obligations for Supplemental Power under the PPC authorized thereby shall constitute the general obligations of the Solar Participant for the payment of which the full faith and credit of the Solar Participant is pledged, obligating the Solar Participant to provide for the assessment and collection of an annual tax sufficient in amount to provide funds annually to make all payments due thereunder; and

WHEREAS, the Mayor and City Council desire to approve the PPC; to authorize the execution and delivery of the PPC and other such documents, certificates, and opinions described therein; and authorize such further actions as necessary for the Solar Participant to procure Supplemental Power as provided thereby.

NOW, THEREFORE, BE IT RESOLVED by the Mayor and City Council of the Solar Participant as follows:

1. <u>Incorporation of Recitals</u>. The recitals set forth above are hereby incorporated in the body of this Resolution.

2. <u>Findings and Determinations</u>. All findings and determinations contained in the PPC, including the recitals thereto, are hereby incorporated herein by reference, and are hereby adopted as findings and determinations of the Mayor and City Council of the Solar Participant.

3. <u>Defined Terms</u>. All capitalized terms used herein but not otherwise defined herein shall have the meanings set forth in the PPC.

4. <u>Supplemental Power Purchase Agreement</u>. The Mayor and City Council of the Solar Participant acknowledge receipt of the form of the SPPA to be executed by the Authority and Company.

5. <u>Authorization to Execute PPC</u>. The Mayor and City Council of the Solar Participant hereby authorize the Solar Participant to enter, as a Solar Participant (defined therein) into the PPC in substantially the form attached as <u>Exhibit A</u> hereto, and to perform the same, and the Mayor of the Solar Participant is hereby authorized on behalf of the Solar Participant to execute and deliver the PPC. The Mayor, with the advice of Counsel to the Solar Participant, is authorized to agree to such changes to the PPC as may be necessary prior to execution thereof, and the execution and delivery of the PPC shall be conclusive evidence of such approval. The City Clerk of the Solar Participant is authorized to attest the execution by the Mayor of the PPC and to affix the seal of the Solar Participant to such documents.

6. Further Authority. The Mayor and City Council hereby authorize, empower and direct the Mayor and any necessary representatives of the Solar Participant to do all such acts and things and to execute all such documents as may be necessary to carry out and comply with the provisions and intent of this Resolution and the PPC.

Authorized Representative. The Mayor and Deputy Mayor of the Solar 7. Participant are each hereby each designated as Authorized Representatives of the Solar Participant, and may execute notices, certificates, requests, estimates and other documents contemplated by the PPC, subject to the limitations contained herein.

Repeal of Conflicting Resolutions. All resolutions and parts of resolutions in conflict with this Resolution are hereby repealed to the extent of such conflict.

9. Effective Date. This Resolution (including the recitals first above written, which are hereby incorporated into this Resolution) shall take effect immediately upon its adoption; a copy of this Resolution may be filed in such offices as the undersigned or such development authority may elect to file this Resolution. All resolutions, or parts of resolutions, in conflict herewith are repealed.

BE IT AND IT IS HEREBY RESOLVED AND ADOPTED BY THE MAYOR AND **CITY COUNCIL OF THE CITY OF CARTERSVILLE (Solar Participant), this** day of _____, 2021.

/s/_____ Matthew J. Santini, Mayor City of Cartersville, Georgia

ATTEST:

/s/____

Julia Drake, City Clerk City of Cartersville, Georgia **EXHIBIT "A" TO RESOLUTION**

POWER PURCHASE CONTRACT BETWEEN MUNICIPAL ELECTRIC AUTHORITY OF GEORGIA AND THE UNDERSIGNED PARTICIPANT

This Power Purchase Contract (this "**Contract**"), made and entered into as of _______, 2021, by and between the Municipal Electric Authority of Georgia (the "**Authority**" or "**MEAG Power**"), a public body corporate and politic and a public corporation and an instrumentality of the State of Georgia, created by the provisions of the Municipal Electric Authority Act, Ga. L. 1976, p. 107, as amended (the "**Act**"), and the City of Cartersville (the "**Solar Participant**"), a political subdivision of the State of Georgia.

WITNESSETH:

WHEREAS, pursuant to the Act, the Authority has previously entered into one or more Power Sales Contracts (each, as amended, a "**Power Sales Contract**") with eligible political subdivisions, including the Solar Participant (each, a "**Participant**") to provide, from defined production projects and sources, for the Participants' bulk electric power supply needs;

WHEREAS, one such Power Sales Contract, the Project One Power Sales Contract (the "**Project One Power Sales Contract**"), further provides in Section 401 thereof that the Authority will provide or cause to be provided to each of the participants thereto, including the Solar Participant, (the "**Project One Participants**") its supplemental bulk power supply ("**Supplemental Power**") (i.e., that portion of the Solar Participant's bulk power supply in excess of its entitlement to power, energy, output and services from any MEAG Power project) during each month of each Power Supply Year (therein defined);

WHEREAS, Section 404 of the Project One Power Sales Contract provides that a Project One Participant may elect to procure an alternate source of Supplemental Power other than that

provided by the Authority, subject to providing notice to the Authority in accordance with subpart (c) of that Section;

WHEREAS, the Authority adopted a Supplemental Power Supply Policy in March of 1999, as amended (the "**Supplemental Power Policy**"), which, in part, waived the notice requirements provided for in Section 404(c) of the Project One Power Sales Contract;

WHEREAS, the Authority has an opportunity to procure a substantial amount of Supplemental Power for a multi-year term through a Power Purchase Agreement with Pineview Solar LLC (the "**Company**") for the output and services of approximately 80 MWac from a photovoltaic solar energy generation facility located in Wilcox County, Georgia (the "**Facility**") to be constructed, owned, operated, and maintained by the Company (hereinafter the "**SPPA**");

WHEREAS, in accordance with the Supplemental Power Policy, the Solar Participant and certain other Project One Participants (each such participating Project One Participant referred to herein as a "**PPOP**" and each such PPOP other than the Solar Participant an "**Other PPOP**") have requested that the Authority purchase from the Company power, output and services of the Facility to provide for their Supplemental Power;

WHEREAS, the Authority and the Solar Participant agree that this Contract is supplemental to and authorized by the Project One Power Sales Contract;

WHEREAS, the Authority has entered into power purchase contracts with the other PPOPs that are substantially similar to this Contract (each such power purchase contract an "**Other PPC**"); provided that each Other PPC reflects the applicable PPOP's Maximum MW Subscription (as defined below);

WHEREAS, the Authority and the Solar Participant agree that the payment obligations under this Contract shall constitute the general obligations of the Solar Participant for the payment

of which the full faith and credit of the Solar Participant is pledged, obligating the Solar Participant to provide for the assessment and collection of an annual tax sufficient in amount to provide funds annually to make all payments due hereunder;

NOW, THEREFORE, for and in consideration of the premises and mutual covenants and agreements hereinafter set forth, it is agreed by and between the parties hereto as follows:

1.

1.1 <u>SPPA</u>. The SPPA, in substantially the form attached hereto as <u>Exhibit A</u>, describes the terms under which the Products (as defined therein) of the Facility shall be made available to the Authority for the provision of solar power to the Solar Participant.

1.2 <u>Entitlement Share</u>.

(a) <u>Maximum MW Subscription</u>: The Solar Participant's "**Maximum MW** Subscription" is _____ MWac.

(b) <u>Entitlement Share</u>. The Solar Participant's "**Entitlement Share**" shall be that percentage of the Facility's output to which the Solar Participant is entitled. The Solar Participant's Entitlement Share shall calculated as follows:

(i) <u>Step One:</u> The amount of the Solar Participant's Maximum MW Subscription shall be multiplied by a fraction, the numerator of which is the number of MWAC actually comprising the Facility and the denominator of which is the sum of the amount of the Solar Participant's Maximum MW Subscription and the amount of the maximum MW subscriptions of the Other PPOPs.

(ii) <u>Step Two</u>: The solution to Step One, above, shall be divided by the number of MWAC actually comprising the Facility (with the solution to this Step Two being the percentage of the Facility's output constituting the Solar Participant's Entitlement Share).

1.3 Initial Payment Obligation. The Authority shall deliver to the Solar Participant an initial billing statement up to ninety (90) days prior to the Facility's anticipated commencement of the delivery of Test Energy pursuant to the SPPA (such anticipated date of delivery referred to as the "Start Date"). The initial billing statement shall set forth the Solar Participant's allocable share of the sum of the estimated Solar Costs and estimated MEAG Costs (both terms, as defined in Section 1.4 below) for the month the Authority anticipates will generate the highest aggregate amount of Solar Costs and MEAG Costs (the "Maximum Monthly Amount") during the year subsequent to the year of the Start Date. Amounts collected pursuant to this Section 1.3 (the "Escrow Amount") shall be held in escrow by the Authority, subject to use by the Authority pursuant to the terms hereof. At the end of each calendar year commencing the year after the year of the Start Date the Authority shall recalculate the Solar Participant's Maximum Monthly Amount for the next year and, (i) if the Maximum Monthly Amount exceeds the Escrow Amount, the Authority shall include an amount equal to such deficit on the Solar Participant's next Billing Statement (as defined in Section 1.4) and (ii) if the Maximum Monthly Amount is less than the Escrow Amount, the Authority shall, at the Authority's election, either (A) refund to the Solar Participant an amount from the Escrow Amount equal to such excess or (B) credit such excess to the Solar Participant's next succeeding Billing Statement(s).

1.4 <u>Ongoing Payment Obligations</u>.

(a) The Authority shall deliver to the Solar Participant a monthly Billing Statement commencing within the thirty (30) days preceding the anticipated Start Date and continuing through the Term. For purposes of this Contract, a "**Billing Statement**" shall be a written statement prepared or caused to be prepared monthly in advance by the Authority that shall set forth the Solar Participant's estimated payment obligations pursuant to the terms hereof.

(b) The Solar Participant shall remit payment monthly in advance. The Solar Participant's payment obligations hereunder for a particular month shall be an amount equal to the Solar Participant's allocable share of the sum of the estimated Solar Costs and the estimated MEAG Costs. To the extent the amount paid by the Solar Participant pursuant to the preceding sentence is either greater or less than the Solar Participant's allocable share of the sum of the actual Solar Costs and the actual MEAG Costs for a particular month, the Authority: (i) shall credit any excess payment to the Solar Participant's next Billing Statement and (ii) may satisfy any deficit from the Solar Participant's Escrow Amount and include a corresponding charge on the Solar Participant's next Billing Statement (so as to restore the Solar Participant's Escrow Amount).

For purposes of this Contract, (i) "Solar Costs" for a particular month shall mean the gross amount due to the Company or any other person for the month by the Authority pursuant to the terms of the SPPA, but excluding any interest charged by the Company to the Authority pursuant to Section 10.3 of the SPPA and (ii) "MEAG Costs" for a particular month shall mean all costs incurred by the Authority during the month in connection with the purchase from the Company and delivery to the Solar Participant of the Solar Participant's Entitlement Share, including, but not limited to, (A) costs of (I) scheduling the delivery of solar energy, (II) energy imbalance penalties and (III) all other charges imposed on the Authority and associated with the transmission and delivery of solar energy to the Solar Participants, and (B) a share determined by the Authority to be allocable to this Contract, of all (I) administrative and general costs and (II) operation and maintenance costs, in each case related to the operation and conducting the business of the Authority, including salaries, fees for legal, engineering, and other services and all other expenses properly related to the conduct of the affairs of the Authority. (c) The Solar Participant's payment obligations to the Authority arising under this Contract shall constitute general obligations of the Solar Participant for the payment of which the full faith and credit of the Solar Participant shall be and the same hereby is pledged to provide the funds required to fulfill all obligations arising under this Contract. Unless such payments or provisions for such payments have been made from the revenues of the Solar Participant's electric system or from other available funds, the Solar Participant will annually in each and every fiscal year during the term of this Contract include in its general revenue or appropriation measure sums sufficient to satisfy the payments required to be made in each year by this Contract until all payments hereunder have been made in full.

(d) Except as specifically provided herein, any payment due under this Contract shall be paid within ten (10) calendar days of the Solar Participant's receipt of the Billing Statement. The Parties agree to work in good faith to resolve any disputed amounts prior to the due date for such amount, and agree that any resolution of such disputed amount may, if necessary be addressed by appropriate adjustment to subsequent Billing Statements.

1.5 <u>Rate Covenant</u>. The Solar Participant will establish, maintain, and collect rates and charges for the electric service of its electric system so as to provide revenues sufficient to enable the Solar Participant to pay to the Authority all amounts payable under this Contract and to pay all other amounts payable from and all lawful charges against or liens on the revenues of its electric system.

2.

<u>Term</u>. The term of this Contract shall commence on the date that is ninety (90) days prior to the Start Date and shall continue through and include the end of the twentieth (20th) Contract Year (as defined in the SPPA), unless the SPPA is terminated prior to such date, at which point

this Contract will terminate upon the Solar Participant's full and complete satisfaction of its duties and obligations hereunder.

3.

<u>Products Constitute Supplemental Bulk Power</u>. The Solar Participant acknowledges that all Products contemplated in the proposed SPPA, if implemented, will constitute Supplemental Power, provided, however, that the Solar Participant agrees that it will not exercise its rights under the Supplemental Power Supply Policy or Section 404(c) of the Project One Power Sales Contract to opt-out of its payment obligations under this Contract at any time prior to the expiration of the term of the SPPA.

4.

<u>Pledge of Payments</u>. All payments in respect of Solar Costs required to be made by the Solar Participant pursuant to this Contract, and any or all rights to collection or enforcement of such payments, may be pledged to secure the payment of the Authority's obligations under the SPPA.

5.

<u>Governing Law; Venue</u>. This Contract shall be interpreted and enforced in accordance with the laws of the State of Georgia, excluding any choice of law rules that may direct the application of the laws of another jurisdiction. The Parties agree that the venue for any action arising out of, or in regard to, this Contract shall be in the Superior Court of Fulton County, Georgia and each Party hereby consents to jurisdiction over it in Fulton County, Georgia.

6.

<u>Mutual Representations and Warranties</u>. Each Party represents and warrants to the other that, as of the Effective Date:

(a) <u>Organization</u>. It is duly organized and validly existing under the laws of the
State of Georgia.

(b) <u>Authority</u>. It (i) has the requisite power and authority to enter into this Contract and (ii) has, or as of the requisite time will have, all regulatory and other authority necessary to perform hereunder.

(c) <u>Corporate Actions</u>. It has taken all corporate or other applicable actions, including provision of notice, required to be taken by it to authorize the execution, delivery and performance hereof and the consummation of the transactions contemplated hereby.

(d) <u>No Contravention</u>. The execution, delivery and performance and observance hereof by it of its obligations hereunder do not (a) contravene any provision of, or constitute a default under, (i) any indenture, mortgage, security instrument or undertaking, or other material agreement to which it is a party or by which it is bound, (ii) any valid order of any court, or any regulatory agency or other body having authority to which it is subject, or (iii) any material Applicable Law presently in effect having applicability to it, or (b) require the consent or approval of, or material filing or registration with, any Governmental Authority or other Person other than such consents or approvals that are not yet required but expected to be obtained in due course.

(e) <u>Valid and Enforceable Agreement</u>. This Contract is a valid and legally binding obligation of it, enforceable against it in accordance with its terms, except as the enforceability hereof may be limited by Georgia law, including the Act, and general principles of equity or bankruptcy, insolvency, bank moratorium or similar laws affecting creditors' rights generally, laws restricting the availability of equitable remedies, and limitations on legal remedies against public bodies corporate and politic of the State of Georgia.

(f) <u>Litigation</u>. No litigation, arbitration, investigation or other proceeding is pending or, to the best of such Party's knowledge, threatened against such Party with respect to this Contract or the transactions contemplated hereunder, in each case, that if it were decided against such Party would materially and adversely affect such Party's ability to perform its obligations hereunder.

(g) <u>Legal Opinions</u>. The Solar Participant shall authorize the execution and delivery of this Contract by resolution of its governing body in substantially the form attached hereto as <u>Exhibit B</u>. Further, the Solar Participant shall deliver to the Authority an opinion of counsel (such counsel to be reasonably acceptable to the Authority) as to the due authorization, execution and delivery and the enforceability of this Contract, in substantially the form attached hereto as <u>Exhibit C</u>.

7.

Default; Remedies for Default.

7.1 <u>Default</u>. Failure of the Solar Participant to timely make to the Authority any of the payments for which provision is made in this Contract shall constitute a default on the part of the Solar Participant (a "**Default**"). A Default may be cured by the Solar Participant's (i) full payment of any past due amounts owed by the Solar Participant to the Authority pursuant to the terms hereof (the "**Primary Cure Payments**"), (ii) full payment of any interest which has accrued thereon (as referenced in Section 7.3, below) (the "**Interest Cure Payments**"), and (iii) with reference to Section 7.8(a), below, full restoration of the Escrow Amount, unless and until the Authority exercises its rights pursuant to Section 7.8(c), below (at which point the Default may no longer be cured).

7.2 <u>Continuing Obligation, Right to Discontinue Service</u>. In the event of a Default, the Solar Participant shall not be relieved of its liability for payment of the amounts in default

(including interest accrued thereon pursuant to Section 7.3, below), and the Authority shall have the right to recover from the Solar Participant any amount in default (including interest accrued thereon pursuant to Section 7.3, below). In enforcement of any such right of recovery, the Authority may bring any suit, action, or proceeding in law or in equity, including mandamus and action for specific performance as may be necessary or appropriate to enforce any covenant, agreement or obligation to make any payment for which provision is made in this Contract against the Solar Participant, and the Authority may, upon the occurrence of a Default and at the Authority's discretion, cease and discontinue providing all or any portion of the Solar Participant's Entitlement Share.

7.3 Interest on Late Payments. Any amounts that are not paid when due hereunder shall bear interest at the Contract Interest Rate from the date due until paid, which rate shall not exceed the maximum permissible under Georgia law. The defaulting Solar Participant shall be and shall remain solely liable for the payment of any interest arising under this Section 7.3. For purposes of this Contract, the "Contract Interest Rate" shall mean one hundred (100) basis points per annum plus the rate per annum equal to the prime lending rate as may from time to time be published in The Wall Street Journal under "Money Rates" on such day (or if not published on such day on the most recent preceding day on which published); provided that if at any time during the Term, the Wall Street Journal no longer publishes a prime lending rate, the prime lending rate for purposes of the calculation of the Contract Interest Rate will be average of the prime interest rates which are announced, from time to time, by the three (3) largest banks (by assets) headquartered in the United States which publish a prime, base or reference rate.

7.4 <u>Levy of Tax for Payment</u>. In the event of a Default, the Solar Participant shall provide for the assessment and collection of an annual tax sufficient in amount to provide funds

annually to make all payments due under the provisions of this Contract in each year over the remainder of the life of this Contract and the Authority shall have the right to bring any suit, action or proceeding in law or in equity, including mandamus and action for specific performance, to enforce the assessment and collection of a continuing direct annual tax upon all the taxable property within the boundaries of the Solar Participant sufficient in amount to provide such funds annually in each year of the remainder of the life of this Contract.

7.5 <u>Other Default by Solar Participant</u>. In the event of a failure of the Solar Participant to establish, maintain, or collect rates or charges adequate to provide revenue sufficient to enable the Solar Participant to pay all amounts due to the Authority under this Contract or in the event of a failure of the Solar Participant to take from the Authority its Supplemental Power in accordance with the provisions of this Contract, or in the event of any default by the Solar Participant under any other covenant, agreement or obligation of this Contract, the Authority may bring any suit, action, or proceeding in law or in equity, including mandamus, injunction and action for specific performance as may be necessary or appropriate to enforce any covenant, agreement or obligation of this Contract against the Solar Participant.

7.6 <u>Default by The Authority</u>. In the event of any default by the Authority under any covenant, agreement or obligation of this Contract, the Solar Participant may bring any suit, action, or proceeding in law or in equity, including mandamus, injunction and action for specific performance as may be necessary or appropriate to enforce any covenant, agreement, or obligation of this Contract against the Authority.

7.7 <u>Abandonment of Remedy</u>. In case any proceeding taken on account of any default shall have been discontinued or abandoned for any reason, the parties to such proceedings shall be restored to their former positions and rights hereunder, respectively, and all rights, remedies, owes,

and duties of the Authority and the Solar Participant shall continue as though no such proceeding had been taken.

7.8 <u>Application of Available Remedies</u>.

(a) In the event of a Default by the Solar Participant pursuant to Section 7.1 hereof, the Authority shall:

(i) Apply the Escrow Amount (as collected from the Solar Participant pursuant to Section 1.3, above) to the defaulting Solar Participant's unpaid obligations hereunder;

(ii) Transfer all or any part of the energy generated by the Facility and attributable to the defaulting Solar Participant's Entitlement Share to other Participants or any other person, firm, association or corporation, public or private (such transferee to be determined at the Authority's discretion), for the fair market value of such energy (a "**Default Sale**"); and

(iii) Use the proceeds of such Default Sale (net of the Authority's expenses incurred to facilitate such Default Sale) to (A) satisfy the balance of the defaulting Solar Participant's unpaid obligations hereunder and/or (B) to the extent such payment obligations have been fully satisfied pursuant to Section 7.8(a)(i) and/or this Section 7.8(a)(iii), fully or partially restore the defaulting Solar Participant's Escrow Amount.

(b) The excess, if any, of the proceeds of the Default Sale (net of the Authority's expenses incurred to facilitate such Default Sale) over the defaulting Solar Participant's unpaid payment obligations for a particular month (calculated pursuant to Section 1.4 and inclusive of any interest amount accrued pursuant to Section 7.3, above) shall be for the benefit of the non-defaulting Other PPOPs.

(c) Notwithstanding any Default Sale, the defaulting Solar Participant shall remain liable to the Authority for the full payment of the amount reflected on its Billing Statements plus any interest accrued thereon as if such Default Sale had not been made; except that such liability shall be discharged by an amount equal to the proceeds of the applicable Default Sale (net of the Authority's expenses incurred to facilitate such Default Sale). In the event the Solar Participant's Default continues uncured for ninety (90) calendar days or the Solar Participant fails to timely satisfy its payment obligations hereunder for either three (3) consecutive months or five (5) out of eight (8) months, the Authority may sell the defaulting Solar Participant's Entitlement Share to the other Participants or any other person, firm, association or corporation, public or private (such transferee to be determined at the Authority's discretion); provided that, if such a transfer occurs, the defaulting Solar Participant shall remain liable to the Authority for the full payment of the amount attributable to its Entitlement Share plus any interest accrued thereon as if such transfer had not been made; except that such liability shall be discharged to the extent that the Authority receives payment (net of the Authority's expenses incurred in facilitating such transfer) from the transferee.

7.9 Obligations with Respect to Defaults of Other PPOPs.

(a) If an Other PPOP (a "**Defaulting PPOP**") defaults on its payment obligations (the amount of such default the "**Default Amount**") pursuant to its Other PPC, then the Authority shall pursue its remedies against such Defaulting PPOP as set forth in Section 7.8(a) of the Defaulting PPOP's Other PPC (which remedies are identical to the provisions set forth in Section 7.8(a) of this Agreement). All of the proceeds generated from the application of such remedies (net of the Authority's expenses incurred in pursuing such remedies) shall be applied to reduce the Default Amount.

(b) The amount of any remaining Default Amount (calculated without including any interest accrued pursuant to Section 7.3 of the Defaulting PPOP's Other PPC) after application of the remedies described in Section 7.9(a), above, is referred to as a "**Special Cost Increase**." Special Cost Increases shall be allocated among the non-defaulting PPOPs (including the Solar Participant) *pro rata* based on their Entitlement Shares. The Solar Participant (along with each other non-defaulting Other PPOP) shall be obligated to satisfy its allocable share of the Special Cost Increase; provided that the Solar Participant's share of a Special Cost Increase shall not exceed 25% of the amount otherwise reflected on the Solar Participant's Billing Statement for the month to which the Special Cost Increase is attributable.

(c) If a Defaulting PPOP cures a default pursuant to Section 7.1 of its Other PPC subsequent to the Solar Participant's (and non-defaulting Other PPOP's) payment of a corresponding Special Cost Increase, then the Authority shall distribute the applicable Primary Cure Payments (as determined pursuant to the Defaulting PPOP's Other PPC) ratably to the nondefaulting PPOPs (including the Solar Participant) who satisfied their ratable share of the Special Cost Increase. Interest Cure Payments attributable to Solar Costs shall be paid by the Authority to the Company in satisfaction of the Authority's obligations under the SPPA. Interest Cure Payments attributable to MEAG Costs shall be distributed to the non-defaulting PPOPs ratably based on their Entitlement Shares.

8.

The Solar Participant shall use commercially reasonable efforts to promptly notify the Authority in writing upon the Solar Participant's receipt of a request for a copy of the SPPA pursuant to the Georgia Open Records Act (O.C.G.A. § 50-14-1, *et seq.*). Such notification shall be provided prior to the Solar Participant's release of the SPPA.

In witness whereof, the Authority has caused this Contract to be executed in its corporate name by its duly authorized officers and the Authority has caused its corporate seal to be hereunto impressed and attested; the Solar Participant has caused this Contract to be executed in its corporate name by its duly authorized officers and its corporate seal to be hereunto impressed and attested, and delivery hereof by the Authority to the Solar Participant is hereby acknowledged, all as of the day and year first above written.

MUNICIPAL ELECTRIC AUTHORITY OF GEORGIA

By:		
Name:	James E. Fuller	
Title:	President and CEO	

ATTEST:

By:	
Name:	
Title:	

(SEAL)

[Solar Participant Signature is on the next page]

CITY OF CARTERSVILLE

By:	
Name:	
Title:	

ATTEST:

By:		
Name:		
Title:		

EXHIBIT A

FORM OF SPPA

[Form of SPPA begins on the next page]

POWER PURCHASE AGREEMENT

BETWEEN

PINEVIEW SOLAR LLC

AND

MUNICIPAL ELECTRIC AUTHORITY OF GEORGIA

dated as of

_____, 2021

TABLE OF CONTENTS (to be updated)

SECTION	I DEFINITIONS; RULES OF INTERPRETATION	2
1.1	Defined Terms	2
1.2	Rules of Interpretation 1	3
SECTION	2 TERM; FACILITY DEVELOPMENT 1	4
2.1	Term1	4
2.2	Facility Construction and Delay Damages and Buydown Liquidated Damages	4
2.3	Damages Calculation 1	4
2.4	Delay Damages Invoicing1	5
2.5	Quarterly and Monthly Reports 1	5
2.6	Tax Credits1	5
2.7	Commercial Operation1	5
SECTION	3 REPRESENTATIONS AND WARRANTIES 1	.6
3.1	Mutual Representations and Warranties1	6
3.2	No Other Representations or Warranties 1	7
SECTION	4 PURCHASE AND SALE OF PRODUCTS1	7
4.1	Purchase and Sale 1	7
4.2	No Sales to Third Parties 1	7
4.3	Title and Risk of Loss 1	7
4.4	Curtailment 1	7
4.5	Environmental Attributes, Capacity Rights and Ancillary Services 1	8
4.6	Other Products	20
SECTION	5 CONTRACT PRICE; COSTS 2	20
5.1	Contract Price	20
5.2	Payment Obligation	20
5.3	Costs and Charges	20
5.4	Taxes	:0
5.5	Costs of Ownership and Operation	21
5.6	Rates Not Subject to Review	21

SECT	ION 6	OPERATION AND CONTROL	22
	6.1	Standard of Facility Operation	22
	6.2	Interconnection	22
	6.3	Coordination with the Transmission Owner	23
	6.4	Outages	23
	6.5	Buyer Transmission Services	24
	6.6	Increase in Actual Capacity After the Final Completion Date	24
	6.7	Meteorological Data	24
	6.8	Forecasting	24
	6.9	Operational Records and Reports	25
	6.10	Production Guarantee	26
	6.11	Access Rights	26
	6.12	Signage	27
SECT	ION 7	GENERATOR STATUS	27
	7.1	Authority to Make Sales	27
SECT	ION 8	SELLER'S SECURITY AND CREDIT SUPPORT	27
	8.1	Seller Credit Support	27
	8.2	Not an Exclusive Remedy	28
SECT	ION 9]	METERING AND COMMUNICATION	28
	9.1	Installation of Metering Equipment	28
	9.2	Metering	28
SECT	ION 10	BILLINGS, COMPUTATIONS AND PAYMENTS	29
	10.1	Monthly Invoices	29
	10.2	Offsets	29
	10.3	Interest on Late Payments	29
	10.4	Disputed Amounts	30
	10.5	Recordkeeping	30
	10.6	Required Records and Audit Rights	30
	10.7	Payment Due Dates	30
	10.8	Rounding	31
SECT	ION 11	DEFAULTS AND REMEDIES	31

1	1.1	Defaults	31
1	1.2	Termination and Remedies	33
1	1.3	Duty to Mitigate	35
1	1.4	Security	35
1	1.5	Cumulative Remedies	35
SECTIO	ON 12	INDEMNIFICATION AND LIABILITY	35
1	2.1	Indemnities	35
1	2.2	Consequential Damages	36
SECTIO	ON 13	INSURANCE	37
1	3.1	Evidence of Insurance	37
1	3.2	Term and Modification of Insurance	37
SECTIO	ON 14	FORCE MAJEURE	37
		Definition of Force Majeure	
1		Suspension of Performance	
1	4.3	Force Majeure Does Not Affect Obligations Already Incurred	38
1	4.4	Strikes	38
1	4.5	Right to Terminate	38
1	4.6	Change in Law	39
SECTIO	ON 15	SEVERAL OBLIGATIONS	39
SECTIO	ON 16	CHOICE OF LAW	39
SECTIO	ON 17	PARTIAL INVALIDITY	39
SECTIO	ON 18	NON-WAIVER	39
SECTIO	ON 19	AUTHORIZATIONS	40
SECTIO	ON 20	SUCCESSORS AND ASSIGNS	40
2	20.1	Restriction on Assignments	40
2	20.2	Binding Nature	40
2	20.3	Permitted Assignments by Seller	40
2	20.4	Permitted Assignments by Buyer	41
SECTIO	ON 21	ENTIRE AGREEMENT	41

SECTION 22	2 NOTICES	41	
22.1	Addresses and Delivery Methods	41	
22.2	Changes of Address		
22.3	Notices to Financing Parties		
SECTION 2.	3 DISPUTE RESOLUTION	42	
23.1	Negotiations		
23.2	Choice of Forum	42	
23.3	Settlement Discussions		
23.4	Waiver of Jury Trial	43	
SECTION 24	SECTION 24 CONFIDENTIALITY		
24.1	Confidential Information; Press Releases	43	
24.2	Treatment of Confidential Information	43	
24.3	Disclosure Required by Law	44	
24.4	Georgia Open Records Act and Georgia Open Meeting Act	44	
24.5	Survival	45	
SECTION 2	5 COUNTERPARTS; ORIGINALS	45	

EXHIBITS

- Exhibit A Description of Facility and Point of Delivery
- Exhibit B Expected Annual Production
- Exhibit C NERC Event Types
- Exhibit D Example Calculation of Buyer's Cost to Cover
- Exhibit E Form of Letter of Credit
- Exhibit F Form of Seller Guaranty
- Exhibit 1 Quarterly/Monthly Reports
- Exhibit 2 Required Insurance

POWER PURCHASE AGREEMENT

THIS POWER PURCHASE AGREEMENT (the "Agreement"), entered into as of this ______ day of ______ 2021, is between Pineview Solar LLC (the "Seller"), a Delaware limited liability company, and the Municipal Electric Authority of Georgia (the "Buyer" or "MEAG Power"), a public body corporate and politic and an instrumentality of the State of Georgia, created by provisions of the Municipal Electric Authority Act, Ga. L. 1976, p. 107, as amended (the "Act"). Seller and Buyer are sometimes hereinafter referred to collectively as the "Parties" and individually as a "Party."

BACKGROUND RECITALS

WHEREAS, Seller intends to construct, own, operate and maintain an 80 MWac nameplate capacity photovoltaic solar energy generation facility located in Wilcox County, Georgia, which is more fully described on **Exhibit A** (the "**Facility**");

WHEREAS, pursuant to the Act, the Buyer has entered into one or more Power Sales Contracts, as amended (each, a "**Power Sales Contract**"), with eligible political subdivisions (each, a "**Participant**"), for the provision of bulk power and other services to such Participants;

WHEREAS, under the applicable Power Sales Contracts, the Buyer has agreed to obtain for or provide to the Participants Supplemental Bulk Power Supply ("**Supplemental Power**");

WHEREAS, the Act authorizes the Buyer to execute power purchase contracts and other agreements with publicly or privately owned entities in order to provide or make available an adequate, dependable, and economical supply of energy and related services to its Participants;

WHEREAS, in accordance with the applicable Power Sales Contracts and the Buyer's Supplemental Power Supply Policy, certain Participants (the "**Solar Participants**") have requested that the Buyer purchase from the Seller power, output and services of the Facility to provide Supplemental Power to the Solar Participants;

WHEREAS, in order to make Supplemental Power generated by the Facility available to the Solar Participants, each Solar Participant has, or will enter into a Power Purchase Contract with the Buyer (each, a "**PPC**") under which such Solar Participant shall obtain its Entitlement Share of the Products (hereinafter defined) and incur its respective share of the payment obligations hereunder; each Solar Participant's payment obligations under its PPC are general obligations to the payment of which its full faith and credit are pledged, obligating such Solar Participant to provide for the assessment and collection of an annual tax sufficient in amount to provide funds annually to make all payments when due thereunder; and

WHEREAS, Seller desires to sell and deliver to Buyer for sale to its Solar Participants, and Buyer desires to purchase and receive from Seller, the Products (as such term is hereinafter defined) associated with the Contract Amount as defined herein, in each case, generated by the Facility, all in accordance with the terms and conditions hereof. NOW, THEREFORE, in consideration of the foregoing and the mutual promises set forth below and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties, intending to be legally bound, mutually agree as follows:

SECTION 1 DEFINITIONS; RULES OF INTERPRETATION

1.1 <u>Defined Terms</u>. Unless otherwise required by the context in which any term appears, initially capitalized terms used herein shall have the following meanings:

"Acceptable Credit Rating" means, with respect to Seller, having a Credit Rating of no less than (as applicable): (a) "BBB-" from S&P, or (b) "Baa3" from Moody's.

"Acceptable REC Registry" means the North American Renewables Registry or any other registry as mutually agreed to by the Parties.

"Act" is defined in the Preamble.

"Actual Capacity" means the maximum installed instantaneous generation capacity of the completed Facility at which the Facility can operate during the first Contract Year of operation, and as adjusted thereafter for degradation, expressed in MWac as measured on the AC side of the inverters, when operated in compliance with the Generation Interconnection Agreement and consistent with the operating parameters provided by the manufacturer of the Facility equipment.

"Affiliate" means, with respect to any entity, each entity that directly or indirectly controls, is controlled by, or is under common control with, such designated entity, with "control" meaning the possession, directly or indirectly, of the power to direct the management and policies, whether through the ownership of voting securities (if applicable) or by contract or otherwise.

"Agreement" is defined in the preamble and includes Exhibits A - F, Exhibits 1 and 2, and any amendments hereto that are executed by the Parties.

"Ancillary Services" means the services associated with the Contract Amount that the Facility is capable of providing, without any modifications to the Facility or the Seller's operation of the Facility, to support the transmission of capacity and energy from generation resources to loads while maintaining the reliable operation of the GITS.

"**Applicable Law**" means any applicable federal, state and local law, statute, regulation, rule, action, order, tariff, code or ordinance enacted, adopted, issued or promulgated by any federal, state, local or other Governmental Authority or regulatory body.

"**Business Day**" means any day Monday through Friday from 9 a.m. to 5 p.m. Eastern Prevailing Time excluding nationally recognized public holidays.

"**Buydown Liquidated Damages**" means an amount equal to the product of (a) the positive difference between (i) the Contract Amount and (ii) the Actual Capacity of the Facility as of the Outside Commercial Operation Date and (b) \$50,000 per MWac. To be clear, if the difference is negative, the Buydown Liquidated Damages is zero dollars (\$0).

"Buyer" is defined in the preamble.

"Buyer Initiated Curtailment" means a scenario in which production and/or deliveries of Net Output from the Facility is curtailed due to an act or omission of Buyer.

"Buyer Indemnitees" is defined in Section 12.1.1.

"Buyer's Cost to Cover" means the product of (a) Buyer's replacement cost for Products (including any charge, cost or expense incurred to import the Products) during the Performance Measurement Period for which the determination is being made, as reasonably determined by Buyer, minus the Contract Price during the Performance Measurement Period, and (b) the Minimum Production Guarantee for such Performance Measurement Period, minus the Net Output for such Performance Measurement Period determined in accordance with Section 6.10.1, which will include all Deemed Delivered Output and Deemed Generated Energy from Curtailments during such Performance Measurement Period (such amount under this clause (b) being the "Deficient Quantity"). If the calculation in clause (a) renders a negative number, then the amount under this clause (a) shall be \$0.00. An Example illustrating the calculation of Buyer's Cost to Cover under certain stated assumptions is set forth in Exhibit D.

"**Capacity Rights**" means any current or future defined characteristic, certificate, benefit, product, tag, credit, attribute, or accounting construct, including any of the same counted towards any current or future resource adequacy or reserve requirements, associated with the electric generation capability and capacity of the Actual Capacity from the Facility, or from the Facility's capability and ability to produce energy. Capacity Rights do not include Environmental Attributes, Ancillary Services or any Tax Credits.

"**Cash Deposit**" means United States currency, deposited with a Qualified Issuer in an interest bearing account, in which Buyer holds a first and exclusive perfected security interest pursuant to an escrow agreement, account control agreement or other agreement, either: (i) in an account under which Buyer is designated as beneficiary with sole authority to withdraw cash from the account or otherwise access the funds in the account; or (ii) held in trust by the Qualified Issuer as escrow agent with instructions to pay claims made by Buyer pursuant to this Agreement.

"Change in Law" is defined in Section 14.6.

"**Commercial Operation**" means that not less than the Required Percentage of the Contract Amount is fully operational and reliable and is fully interconnected and synchronized with the GITS, which occurs when all of the following events have occurred:

a. Seller shall have entered into a Generation Interconnection Agreement with Transmission Owner;

b. Buyer shall have received a certificate addressed to Buyer from an officer of Seller certifying that: (i) the Nameplate Capacity of the Facility is at least equal to the Required Percentage of the Contract Amount; (ii) the Facility is able to generate electric power reliably in accordance with the terms and conditions of this Agreement and the Generation Interconnection Agreement; and (iii) all Permits to construct and/or operate the Facility in compliance with all

Applicable Law and this Agreement have been obtained and are in full force and effect, other than Permits which would not adversely affect Seller's ability to operate the Facility;

c. Buyer shall have received a certificate addressed to Buyer from a Licensed Professional Engineer certifying that: (i) in accordance with the Generation Interconnection Agreement, all required Interconnection Facilities have been constructed; (ii) all required interconnection tests have been completed; and (iii) not less than the Required Percentage of the Contract Amount is physically interconnected with the GITS in accordance with the Generation Interconnection Agreement and synchronized with the GITS;

d. Seller has provided the Seller Credit Support to Buyer as required pursuant to Section 8.1; and

e. No Seller Event of Default is outstanding or remains uncured in accordance with Article 11.

"**Commercial Operation Date**" means the date that Commercial Operation is achieved or deemed achieved pursuant to Section 2.7.

"Confidential Information" is defined in Section 24.1.

"**Construction Credit Support**" means a Letter of Credit, Cash Deposit, or Guaranty, or a combination thereof, provided by Seller for the benefit of Buyer in an amount equal to Three Million Dollars (\$3,000,000).

"Contract Amount" means 80 MWac. In the case that Buydown Liquidated Damages are paid by Seller, the Contract Amount shall be adjusted as of the Outside Commercial Operation Date pursuant to Section 2.2.3.

"Contract Interest Rate" means one hundred (100) basis points per annum plus the rate per annum equal to the prime lending rate as may from time to time be published in The Wall Street Journal under "Money Rates" on such day (or if not published on such day on the most recent preceding day on which published); provided that if at any time during the Term, the Wall Street Journal no longer publishes a prime lending rate, the prime lending rate for purposes of the calculation of the Contract Interest Rate will be average of the prime interest rates which are announced, from time to time, by the three (3) largest banks (by assets) headquartered in the United States which publish a prime, base or reference rate.

"Contract Price" means \$25.91 per MWh.

"Contract Year" means each consecutive twelve (12) month period during the Term that commences with the Commercial Operation Date or one of its anniversaries, provided that if this Agreement is terminated prior to its expiration, the Contract Year in which termination occurs will begin on the anniversary of the Commercial Operation Date immediately preceding the termination date (or if such termination occurs during the first Contract Year, on the Commercial Operation Date) and will end on the date of the termination of this Agreement. "**Costs**" means, with respect to the non-defaulting Party, brokerage fees, commissions and other similar third party transaction costs and expenses reasonably and actually incurred by such Party either in terminating any arrangement pursuant to which it has hedged its obligations (which, for the sake of clarity, does not include the non-defaulting Party's losses or gains with respect to any such hedging arrangement) or entering into new arrangements which replace this Agreement (including costs incurred in connection with transmission services that would otherwise not have been incurred hereunder), and to the extent permitted by Georgia law, all reasonable attorneys' fees and expenses incurred by the non-defaulting Party in connection with remedies initiated pursuant to the provisions of this Agreement.

"Credit Rating" means the rating then assigned to Seller's senior, unsecured long-term debt obligations (not supported by third party credit enhancements) or if Seller does not have a rating for its senior, unsecured long-term debt, then the rating assigned to Seller as an issuer rating by S&P, Moody's or any other rating agency agreed to by Buyer.

"Deemed Delivered Output" is defined in Section 4.4.2.

"Deemed Generated Energy from Curtailments" is defined in Section 4.4.1.

"**Defaulting Solar Participant**" means a Solar Participant that fails to timely make payments to Buyer pursuant to the terms of its PPC.

"Deficient Quantity" is defined in the definition of "Buyer's Cost to Cover."

"**Delay Damages**" means the damages payable by Seller, under the circumstances and subject to the limits described in Sections 2.2.1 or 2.2.2, which for any given day are equal to the product of Fifty Dollars (\$50) per MWac and the Nameplate Capacity of the Facility required to deliver the Contract Amount.

"**DSP Failure**" means the default listed in Sections 11.1.3(a)(2).

"Effective Date" is defined in Section 2.1.

"**Electric System Authority**" means each of NERC, a regional or sub-regional reliability council or authority, and any other similar council, corporation, organization or body of recognized standing with respect to the operations of the electric system in the geographic area in which the Facility is located.

"**Entitlement Share**" means the portion, measured in MWac, of the Contract Amount to which a Solar Participant is entitled pursuant to its PPC.

"**Entitlement Share Percentage**" means a percentage, calculated with respect to each Solar Participant by dividing such Solar Participant's Entitlement Share by the Contract Amount. The sum of each Solar Participant's Entitlement Share Percentage shall equal 100%.

"**Emergency**" means an abnormal GITS condition requiring manual or automatic action to maintain GITS frequency, or to prevent loss of firm load, equipment damage, or tripping of system

elements that could adversely affect the reliability of an electric system or the safety of persons or property or a condition that requires implementation of emergency procedures.

"Environmental Attributes" means any and all current and future attributes, claims, credits, benefits, emissions reductions, offsets, and allowances, howsoever entitled that are created or otherwise arise from the existence, ownership, or operation of the Facility associated with the Net Output, resulting from the avoidance of the emission of any gas, chemical, or other substance to the air, soil or water. Environmental Attributes include, but are not limited to, the following (a) RECs, (b) any avoided emissions of pollutants to the air, soil, or water such as (subject to the foregoing) sulfur oxides, nitrogen oxides, carbon monoxide, and other pollutants; and (c) any avoided emissions of carbon dioxide, methane, and other greenhouse gases as defined by U.S. laws or regulations as of the Effective Date or as they may be modified during the Term. Environmental Attributes do not include (i) the Tax Credits or any local, state or federal cash grants, depreciation deductions or other tax credits providing a tax benefit to Seller or any other Person based on ownership of, or energy production from, any portion of the Facility or (ii) cash grants, depreciation deductions and other tax benefits arising from ownership or operation of the Facility. In the case of each of the foregoing clauses (i) and (ii), as between the Parties, Seller shall maintain all rights, title and interest in and to such items.

"Event of Default" is defined in Section 11.1.

"**Example**" means an example of certain calculations to be made hereunder. Each Example is for purposes of illustration only and is not intended to constitute a representation, warranty or covenant concerning the matters assumed for purposes of each Example.

"Expected Annual Production" means the expected production from the Contract Amount for each Contract Year as set forth in Exhibit B.

"Facility" is defined in the Recitals and is more fully described in attached Exhibit A.

"FERC" means the Federal Energy Regulatory Commission or its successor.

"**Final Completion Date**" means the earlier of (a) the Date on which the Facility achieves Commercial Operation with respect to one hundred percent (100%) of the Contract Amount required to deliver the Contract Amount or (b) if Buydown Liquidated Damages are paid pursuant to Section 2.2.3, the Outside Commercial Operation Date.

"Financing Party" means any Person other than a Seller Affiliate (or any trustee or agent on behalf of such Person) lending money or extending credit (including any financing lease, monetization of tax benefits, transaction with a tax equity or cash equity investor, back-leverage financing or credit derivative arrangement) to Seller or its Affiliates, all or a portion of which is used (a) for the construction, term or permanent funding, financing or refinancing of the Facility; (b) for working capital or other ordinary business requirements for the Facility (including for the maintenance, repair, replacement or improvement of the Facility); or (c) for any development funding, financing, bridge financing, credit support, credit enhancement or interest rate protection in connection with the Facility. "Financing Party Consent or Estoppel" means (a) a collateral assignment consent agreement, to be entered into by Seller, Buyer and Seller's Financing Parties, in a form reasonably agreed by such parties and containing customary terms and conditions, that recognizes and consents to (i) Seller's collateral assignment of rights and obligations under this Agreement and (ii) the Financing Parties' rights to be notified of, and allowed to cure, any breach or default of this Agreement by Seller, and to exercise, subject to the notice, cure rights, and remedies provisions of this Agreement, any step-in rights consented to by Seller, and other customary terms as reasonably may be requested by such Financing Parties or (b) other agreements with Financing Parties reasonably requested by such Financing Parties, containing customary terms and conditions, including an estoppel certificate or other agreements with Financing Parties reasonably requested by such Financing Parties, in a form reasonably agreed by such parties and containing customary terms and containing customary terms and conditions, including an estoppel certificate or other agreements with Financing Parties reasonably requested by such Financing Parties, in a form reasonably agreed by such parties and containing customary terms and conditions for tax equity or cash equity investors.

"Force Majeure" is defined in Section 14.1.

"Forced Outage" means any unplanned outage or derating of the Facility that results in the reduction of, cessation in the delivery of, or inability to deliver the Products, and specifically includes NERC Event Types U1, U2 and U3, as set forth in attached **Exhibit C** (or similar successor unplanned outages if NERC terminology or concepts are revised during the Term of this Agreement), and specifically excludes any Maintenance Outage or Planned Outage.

"GAAP" means generally accepted accounting principles in the United States of America as in effect from time to time.

"Gains" means, with respect to a non-defaulting Party, an amount equal to the present value of the economic benefit to it, if any (exclusive of Costs), resulting from the termination of this Agreement, determined in a commercially reasonable manner. Gains shall be measured on the basis of one hundred percent (100%) of the Expected Annual Production for each Contract Year (or portion thereof) during the remainder of the Term (ignoring any early termination of this Agreement).

"Generation Interconnection Agreement" means that certain Large Generator Interconnection Agreement by and between Seller and Transmission Owner, and any amendments thereto, containing terms and conditions governing the interconnection and parallel operation of the Facility with GITS.

"Georgia Integrated Transmission System (GITS)" means that statewide transmission system jointly owned by MEAG Power, Georgia Transmission Corporation, Dalton Utilities and Georgia Power. Individual transmission lines and substations that make up the GITS are owned and maintained by the individual participants but they are planned and operated as one system, providing each owner with transmission access throughout Georgia.

"Good Industry Practice(s)" means any applicable practices, methods, and acts engaged in or approved by a significant portion of the electric industry for construction, interconnection and operation of facilities similar to the Facility during the relevant time period, or the practices, methods and acts which, in the exercise of reasonable judgment by a prudent solar energy generation operator in light of the facts known or which should reasonably have been known at the time the decision was made, could have been expected to accomplish the desired result consistent with good business practices, manufacturers' recommendations, reliability, safety, expedition, Applicable Law and the requirements of any Governmental Authority having jurisdiction. "Good Industry Practices" is not intended to be limited to the optimum practice, method or act to the exclusion of all others, but rather is intended to include the acceptable practices, methods or acts generally accepted by the photovoltaic solar generating industry for facilities similar to the Facility during the relevant time period.

"Governmental Authority" means any supranational, federal or state authority or other political subdivision thereof, having jurisdiction over Seller, Buyer, the Facility or this Agreement, including any municipality, township or county, and any entity or body exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to government, including any corporation or other entity owned or controlled by any of the foregoing. An Electric System Authority shall also be considered to be a Governmental Authority.

"Green-e Renewable Energy Standard" means the Green-e Renewable Energy Standard Version 3.3, or successor version.

"Guaranteed Commercial Operation Date" means December 31, 2023, provided that the Guaranteed Commercial Operation Date shall be extended on a day-for-day basis for each day of delay in Seller's development, permitting, construction, interconnection or completion of the Facility associated with (a) the occurrence of a Force Majeure event, (b) a breach by Buyer of any of its obligations under this Agreement, (c) the occurrence of an Emergency condition, or (d) a delay in the in-service date of the Interconnection Facilities beyond the expected date set forth in the Generation Interconnection Agreement, including as a result of a delay in the completion of any Network Upgrades, provided that such delay is not the result of Seller's failure to perform its obligations under the Generation Interconnection Agreement.

"Guarantor" means a Person that has issued a Guaranty and (a) has an Acceptable Credit Rating or (b) has otherwise been approved by Buyer, in Buyer's reasonable discretion, in the case of a Guaranty that is provided as Seller Credit Support.

"Guaranty" means a guaranty of payment (and not performance) issued by a Guarantor substantially in the form attached hereto as **Exhibit F-1** or otherwise in form and substance satisfactory to Buyer, in Buyer's reasonable discretion; it being acknowledged and agreed that, in the event Seller is obligated to post Seller Credit Support pursuant to the terms of this Agreement and Buyer is not satisfied with the form and substance of the proposed Guaranty (in Buyer's reasonable discretion), the Seller shall nevertheless be obligated to post the required Seller Credit Support by means of a Letter of Credit or Cash Deposit.

"Interconnection Facilities" means all the facilities installed, or to be installed, for the purpose of interconnecting the Facility to the GITS, including electrical transmission lines, line upgrades, transformers, capacitor banks, inductor banks, metering, telecommunications, and associated equipment, substations, relay and switching equipment, and safety equipment, including any Network Upgrades.

"Internal Revenue Code" means the Internal Revenue Code of 1986, as such law may be amended or superseded.

"Involuntary Curtailment" is defined in Section 4.4.1.

"**ITCs**" means the investment tax credits established pursuant to Section 48 of the Internal Revenue Code, as such law may be amended or superseded.

"**kWh**" means kilowatt hour.

"Letter of Credit" means an unconditional, irrevocable letter of credit issued by a Qualified Issuer either (a) substantially in the form of the letter of credit attached hereto as **Exhibit E** or (b) otherwise in form and substance satisfactory to Buyer, in Buyer's reasonable discretion, in the case of a Letter of Credit provided as Seller Credit Support

"Liabilities" is defined in Section 12.1.1.

"Licensed Professional Engineer" means a Person proposed by Seller and acceptable to Buyer in its reasonable judgment who (a) is licensed to practice engineering in the United States, and in all States for which such Person is providing a certification, evaluation or opinion with respect to matters specific to such State, (b) has no economic relationship or association with Seller or Buyer other than services previously or currently being rendered to Seller or Buyer or their respective Affiliates in a capacity similar to the services being provided by such Person in relation to the Facility, and (c) is not a representative of Seller or Buyer or a manufacturer or supplier of any equipment installed in the Facility.

"Losses" means, with respect to a non-defaulting Party, an amount equal to the present value of the economic loss to it, if any (exclusive of Costs), resulting from termination of this Agreement, determined in a commercially reasonable manner. Losses shall be measured on the basis of one hundred percent (100%) of the Expected Annual Production for each Contract Year (or portion thereof) during the remainder of the Term (ignoring any early termination of this Agreement).

"Maintenance Outage" means any planned outage of the Facility for maintenance that occurs before the next regularly scheduled Planned Outage, which results in reduction of, cessation in the delivery of, or inability to deliver, the energy associated with the Actual Capacity to the Point of Delivery, and specifically includes NERC Event Type MO, as set forth in attached **Exhibit C** (or similar successor planned maintenance outages if NERC terminology or concepts are revised during the Term of this Agreement), and specifically excludes a Forced Outage or a Planned Outage.

"**MEAG Power**" is defined in the Preamble.

"Minimum Production Guarantee" is defined in Section 6.10.1.

"Moody's" means Moody's Investors Service, Inc. or its successor.

"**MW**" means megawatt.

"**MWac**" means megawatts (associated with output, rating or capacity of a solar PV facility) as measured after conversion of the direct current into alternating current by the Facility invertors.

"**MWh**" means megawatt hour.

"**Nameplate Capacity**" means, as of any date, the aggregate installed nameplate capacity of the Facility, expressed in MWac, as of such date.

"NERC" means the North American Electric Reliability Corporation or its successor.

"**Net Output**" means, for any period, the amount of energy generated by the Facility and delivered to the Point of Delivery, as measured by the Revenue Meter (it being understood that all electrical energy produced by the Facility, excluding line losses and Station Use, shall be delivered to the Point of Delivery).

"**Network Upgrades**" means any upgrades to the GITS that are required in order to provide interconnection service for the Facility and for which the Facility is allocated costs under the Generation Interconnection Agreement or any affected system agreement with an adjoining electric transmission system owner or operator.

"Notice to Proceed Date" means the date on which notice is issued by Seller to its contractor under the engineering, procurement and construction agreement or similar contract relating to the construction of the Facility, authorizing and directing the full and unrestricted commencement of construction of the Facility. The Notice to Proceed Date shall occur on or before October 1, 2022.

"Output" means all energy produced by the Facility.

"Outside COD Termination Payment" is defined in Section 2.2.2.

"**Outside Commercial Operation Date**" means the date that is three hundred and sixty-five (365) days after the Guaranteed Commercial Operation Date.

"**Participant**" is defined in the Preamble.

"Party" and "Parties" are defined in the Preamble.

"Performance Measurement Period" means the full one (1) Contract Year period commencing on the first day of the first Contract Year and the full one (1) Contract Year period commencing on each subsequent Contract Year.

"**Permits**" means all of the permits, licenses, approvals, certificates, entitlements and other authorizations issued by Governmental Authorities required for the construction, ownership or operation of the Facility, sale and delivery of Net Output, and occupancy of the Premises, and all amendments, modifications, supplements, general conditions and addenda thereto.
"**Person**" means any individual, corporation, company, voluntary association, partnership, joint venture, trust, limited liability company, unincorporated organization or Governmental Authority.

"Planned Outage" means any regularly scheduled planned outage of the Facility for maintenance, repair or other purposes, which results in the reduction of, cessation in the delivery of, or inability to deliver energy associated with more than ten percent (10%) of the Actual Capacity of the Facility to the Point of Delivery, and specifically includes NERC Event Type PO, as set forth in attached **Exhibit C** (or similar successor planned outages, if NERC terminology or concepts are revised during the Term of this Agreement), and specifically excludes any Maintenance Outage or Forced Outage.

"**Point of Delivery**" means the point of interconnection between the Facility and the GITS as specified in the Generation Interconnection Agreement, and is described in attached **Exhibit A**.

"Power Sales Contract" is defined in the Preamble.

"**PPC**" is defined in the Preamble.

"Pre-Construction Credit Support" means a Letter of Credit, Cash Deposit, Guaranty, or a combination thereof, as determined by Seller, provided by Seller for the benefit of Buyer in an amount equal to One and One Half Million Dollars (\$1,500,000).

"**Premises**" means the real property on which the Facility is or will be located, as more fully described on **Exhibit A**.

"**Prevailing Time**" or "**PT**" means Eastern Standard Time or Eastern Daylight Time, as applicable on the day in question.

"**Products**" means the Net Output, Environmental Attributes, Capacity Rights and Ancillary Services.

"Qualified Issuer" means a U.S. commercial bank (or a foreign bank with a U.S. branch) authorized under Applicable law to perform the actions described in this Agreement for the benefit of the Buyer and having total assets of at least ten billion dollars (\$10,000,000,000) and a Credit Rating of no less than (as applicable): (a) "A-" from S&P, or (b) "A3" from Moody's, or (c) if such bank has a Credit Rating at such time from both S&P and Moody's, "A-" from S&P and "A3" from Moody's.

"**REC**" or "**Renewable Energy Credit**" shall mean tradable renewable energy credits that contain all the greenhouse gas (GHG) emissions reduction benefits, including carbon dioxide (CO2) reduction benefits, associated with the MWh of renewable electricity when it was generated and are therefore eligible to be certified (whether or not actually certified) in accordance with the Green-e Renewable Energy Standard.

"**Required Facility Documents**" means all Permits, authorizations, rights and agreements reasonably necessary for construction, operation, and maintenance of the Facility.

"Required Percentage" means ninety-five percent (95%).

"**Revenue Meter**" means the meter installed at the Point of Delivery in order to measure the energy delivered by the Facility.

"**ROFR Notice**" is defined in Section 6.6.

"S&P" means Standard & Poor's Financial Services, LLC (a subsidiary of The McGraw-Hill Companies, Inc.) or its successor.

"Seller" is defined in the preamble.

"Seller Credit Support" means (a) for the period commencing ten (10) Business Days after the Effective Date and ending ten (10) days after the issuance of the Notice to Proceed, the Pre-Construction Credit Support, (b) for the period commencing ten (10) days after the issuance of the Notice to Proceed and ending on the day before the Commercial Operation Date, the Construction Credit Support, (c) for the period commencing on the Commercial Operation Date and ending on the last day before the tenth (10th) Contract Year (or, if earlier, the termination of this Agreement and the payment of all amounts owed to Buyer hereunder), the Subsequent Credit Support Tier I, and (d) for the period commencing on the first day of the 11th Contract Year and lasting until the expiration or termination of this Agreement and the payment of all amounts owed to Buyer hereunder, the Subsequent Credit Support Tier II.

"Seller Indemnitees" is defined in Section 12.1.2.

"Solar Participant(s)" is defined in the Preamble and is further defined to mean each of Buyer's Participants that hold an Entitlement Share to the Products pursuant to a PPC between the Solar Participant and Buyer. The Solar Participants and their respective Entitlement Share Percentages, are set forth on <u>Schedule 1</u>, attached hereto.

"**Station Use**" means energy produced by the Facility used to operate the Facility or to perform preventative or corrective maintenance to the Facility.

"**Subsequent Credit Support Tier I**" means a Letter of Credit, Cash Deposit, Guaranty, or a combination thereof, provided by Seller for the benefit of Buyer in an amount equal to Four Million Dollars (\$4,000,000).

"**Subsequent Credit Support Tier II**" means a Letter of Credit, Cash Deposit, Guaranty, or a combination thereof, provided by Seller for the benefit of Buyer in an amount equal to Three Million Dollars (\$3,000,000).

"Supplemental Power" is defined in the Background Recitals.

"Supplemental Power Supply Policy" means that certain policy regarding Supplemental Power Supply adopted by Buyer on March 17, 1999, as revised and adopted on October 20, 1999, and as further revised and adopted on September 19, 2002.

"**Tax Credits**" means any state, local and/or federal production tax credit, tax deduction, and/or investment tax credit specific to the production of renewable energy and/or investments in renewable energy facilities, including ITCs.

"Term" is defined in Section 2.1.

"**Termination Payment**" means, with respect to a Party, the positive difference, if any, between (a) Losses and Costs incurred by such Party as a result of termination of this Agreement, less (b) Gains of such Party as a result of termination of this Agreement, expressed in U.S. dollars. If the Termination Payment calculation results in a number less than or equal to zero dollars (\$0), the Termination Payment shall be zero dollars (\$0).

"**Test Energy**" means any Net Output during periods prior to the Commercial Operation Date and any associated Capacity Rights.

"Transmission Owner" means MEAG Power or any successor that owns the transmission lines, Interconnection Facilities (other than those Interconnection Facilities owned by Seller) and other equipment and facilities with which the Facility interconnects at the Point of Delivery.

1.2 <u>Rules of Interpretation</u>.

1.2.1 General. Terms used in this Agreement but not specifically defined in this Section 1 shall have meanings as commonly used in the English language and, where applicable, in Good Industry Practices. Words not otherwise defined herein that have well known and generally accepted technical or trade meanings are used in accordance with such recognized meanings. Unless otherwise required by the context in which any term appears, (a) the singular includes the plural and vice versa; (b) references to "Articles," "Sections," "Schedules," "Annexes," "Appendices" or "Exhibits" are to articles, sections, schedules, annexes, appendices or exhibits hereof; (c) all references to a particular entity include a reference to such entity's successors; (d) "herein," "hereof" and "hereunder" refer to this Agreement as a whole; I all accounting terms not specifically defined herein shall be construed in accordance with GAAP consistently applied; (f) the masculine includes the feminine and neuter and vice versa; (g) "including" means "including, without limitation" or "including, but not limited to"; and (h) all references to a particular law or statute means that law or statute, as amended from time to time.

1.2.2 Terms Not to be Construed For or Against Either Party. Each term hereof shall be construed simply according to its fair meaning and not strictly for or against either Party. The Parties have jointly prepared this Agreement, and no term hereof shall be construed against a Party on the ground that the Party is the author of that provision.

1.2.3 Headings. The headings used for the sections and articles hereof are for convenience and reference purposes only and shall in no way affect the meaning or interpretation of the provisions hereof.

1.2.4 Examples. Example calculations and other Examples set forth herein are for purposes of illustration only and are not intended to constitute a representation, warranty or covenant concerning the Example itself or the matters assumed for purposes of such Example. If there is a conflict between an Example and the text hereof, the text shall control.

1.2.5 Agreement is a Service Contract. The Parties acknowledge and agree that this Agreement is a service contract within the meaning of Section 7701(e) of the Internal Revenue Code.

SECTION 2 TERM; FACILITY DEVELOPMENT

2.1 <u>Term</u>. This Agreement is entered into as of the date hereof (the "Effective Date") and, unless earlier terminated as provided herein, shall remain in effect until the end of the twentieth (20th) Contract Year (the "Term").

2.2 Facility Construction and Delay Damages and Buydown Liquidated Damages.

2.2.1 Seller shall use commercially reasonable efforts to achieve Commercial Operation on or before the Guaranteed Commercial Operation Date. Seller shall pay Delay Damages to Buyer for each day after the Guaranteed Commercial Operation Date through the Commercial Operation Date unless the Agreement is terminated pursuant to Section 2.2.2.

2.2.2 If the Commercial Operation Date has not been achieved on or before the Outside Commercial Operation Date, Buyer may terminate this Agreement by written notice to other Party on or before the tenth (10th) day following the Outside Commercial Operation Date. If Buyer elects to terminate this Agreement pursuant to this Section 2.2.2, (a) this Agreement shall terminate and thereafter have no force or effect, (b) Seller shall owe to Buyer liquidated damages equal to the amount of the Construction Credit Support required as of such date, reduced by the aggregate amount of all damages or payments (including Delay Damages) paid by Seller to Buyer on or before such date (the "Outside COD Termination Payment"), and (c) neither Party shall have any further obligations or liabilities hereunder.

2.2.3 If Commercial Operation is declared before the Facility is capable of delivering the full Contract Amount, Seller shall use commercially reasonable efforts to cause the Facility to reach the capability of delivering the full Contract Amount. If the Facility has not reached the capability of delivering the full Contract Amount on or before the Outside Commercial Operation Date, Seller shall pay to Buyer liquidated damages in the amount of the Buydown Liquidated Damages, in which case the Contract Amount shall, for the remainder of the Term, be equivalent to the Actual Capacity of the Facility as of the Outside Commercial Operation Date.

2.3 <u>Damages Calculation</u>. Each Party agrees and acknowledges that (a) the damages that Buyer would incur due to Seller's failure to achieve Commercial Operation by the Guaranteed Commercial Operation Date or due to the Actual Capacity of the Facility being less than that required to deliver the full Contract Amount as of the Outside Commercial Operation Date would be difficult or impossible to predict with certainty, and (b) it is impractical and difficult to assess actual damages in the circumstances stated, and therefore the Delay Damages, Outside COD Termination Payment, and Buydown Liquidated Damages. Delay Damages, the Outside COD Termination Payment, and Buydown Liquidated Damages shall be payable in lieu of actual damages and, notwithstanding any other provision of this Agreement:

(i) the Buyer's right to terminate this Agreement and receive the Outside COD Termination Payment pursuant to Section 2.2.2 shall be Buyer's sole and exclusive remedy and Seller's sole and exclusive liability for any failure to achieve Commercial Operation by the Outside Commercial Operation Date. For the avoidance of doubt, such failure shall not be considered an Event of Default; and

(ii) Buydown Liquidated Damages shall be Buyer's sole and exclusive remedy and Seller's sole and exclusive liability for the Actual Capacity of the Facility being less than the Contract Amount as of the Outside Commercial Operation Date. For the avoidance of doubt, such failure shall not be considered an Event of Default.

2.4 <u>Delay Damages Invoicing</u>. By the fifteenth (15th) day following the end of a calendar month in which any Delay Damages have accrued, Buyer shall deliver to Seller an invoice showing Buyer's computation of Delay Damages and any amount due Buyer in respect thereof for the preceding calendar month. No later than thirty (30) days after issuance of such an invoice, Seller shall pay to Buyer, by wire transfer of immediately available funds to an account specified in writing by Buyer or by any other means agreed to by the Parties in writing from time to time, the undisputed amount set forth as due in such invoice.

2.5 <u>Quarterly and Monthly Reports</u>. From the Effective Date until thirty (30) days following the Final Completion Date:

(a) On or before the fifteenth (15th) day of each calendar quarter prior to the Notice to Proceed Date, commencing with the first full calendar quarter beginning after the Effective Date, Seller shall provide a quarterly status report containing the information set forth in Exhibit 1; and

(b) On or before the fifteenth (15th) day of each calendar month commencing with the calendar month next following the calendar month in which the Notice to Proceed Date occurs, Seller shall provide a monthly status report containing the information set forth in Exhibit 1.

Upon Buyer's reasonable request, representatives of Seller and Buyer shall schedule and attend telephone or in person conferences periodically during such period to discuss the status of (i) the development, construction and installation of the Facility and (ii) the achievement of Commercial Operation. Also, upon Buyer's request, Seller will facilitate Buyer's onsite inspection and monitoring of construction activities in accordance with and subject to Section 6.11.

2.6 <u>Tax Credits</u>. Seller shall bear all risks, financial and otherwise, throughout the Term, associated with Seller's or the Facility's eligibility to receive ITCs or other Tax Credits, or to qualify for accelerated depreciation for Seller's accounting, reporting or tax purposes.

2.7 <u>Commercial Operation</u>. Seller shall provide written notice to Buyer stating when Seller believes that Commercial Operation has been achieved and the Actual Capacity of the Facility that will be provided to Buyer at such time together with (a) if applicable, Seller's statement of the Actual Capacity of the Facility that Seller intends to complete but that is not yet completed and (b) the certificates described in the definition of Commercial Operation. Buyer shall have ten (10) Business Days after receipt of such notice to state with specificity any requirements that Buyer reasonably believes have not been satisfied. If, within such ten (10) Business Day period, Buyer does not respond or Buyer notifies Seller confirming that Commercial Operation has been achieved, the original date of receipt of Seller's written notice shall be the Commercial Operation Date. If Buyer notifies Seller within such five (10) Business Day period that Buyer believes Commercial Operation has not been achieved, Seller shall address the concerns stated in Buyer's notice, and Commercial Operation shall occur on the date such concerns are addressed to the mutual satisfaction of both Parties, as specified in a notice from Buyer to Seller, or as otherwise determined pursuant to the dispute resolution provisions set forth in Section 23. The Parties agree that review and approval of the conditions to Commercial Operation may occur on an ongoing and incremental basis as such conditions are satisfied.

SECTION 3 REPRESENTATIONS AND WARRANTIES

3.1 <u>Mutual Representations and Warranties</u>. Each Party represents and warrants to the other that, as of the Effective Date:

3.1.1 Organization. It is duly organized and validly existing under the laws of the jurisdiction of its organization.

3.1.2 Authority. It (a) has the requisite power and authority to enter into this Agreement, (b) has, or as of the requisite time will have, all regulatory and other authority necessary to perform hereunder, and (c) is duly qualified and in good standing under the laws of each jurisdiction where its ownership, lease or operation of property or the conduct of its business requires such qualification.

3.1.3 Corporate Actions. It has taken all corporate, limited liability company or other applicable actions, including provision of notice, required to be taken by it to authorize the execution, delivery and performance hereof and the consummation of the transactions contemplated hereby.

3.1.4 No Contravention. The execution, delivery and performance and observance hereof by it of its obligations hereunder do not (a) contravene any provision of, or constitute a default under, (i) any indenture, mortgage, security instrument or undertaking, or other material agreement to which it is a party or by which it is bound, (ii) any valid order of any court, or any regulatory agency or other body having authority to which it is subject, or (iii) any material Applicable Law presently in effect having applicability to it, or (b) require the consent or approval of, or material filing or registration with, any Governmental Authority or other Person other than such consents or approvals that are not yet required but expected to be obtained in due course.

3.1.5 Valid and Enforceable Agreement. This Agreement is a valid and legally binding obligation of it, enforceable against it in accordance with its terms, except as the enforceability hereof may be limited by Georgia law, including the Act, and general principles of equity or bankruptcy, insolvency, bank moratorium or similar laws affecting creditors' rights generally, laws restricting the availability of equitable remedies, and limitations on legal remedies against public bodies corporate and politic of the State of Georgia.

3.1.6 Litigation. No litigation, arbitration, investigation or other proceeding is pending or, to the best of such Party's knowledge, threatened against such Party or any Affiliate of such Party with respect to this Agreement or the transactions contemplated hereunder, in each case, that if it were decided against such Party would materially and adversely affect such Party's ability to perform its obligations hereunder.

3.1.7 Service Contract. The Parties agree and acknowledge this Agreement purports to be a "service contract" within the meaning of Section 7701I of the Internal Revenue Code, the Parties hereto intend it to be such, and the Agreement should be construed accordingly.

3.2 <u>No Other Representations or Warranties</u>. Each Party acknowledges that it has entered hereinto in reliance upon the representations and warranties set forth in this Agreement, and that no other representations or warranties have been made by the other Party with respect to the subject matter hereof.

SECTION 4 PURCHASE AND SALE OF PRODUCTS

4.1 <u>Purchase and Sale</u>. Except as otherwise expressly provided herein, commencing on the Commercial Operation Date and continuing through the Term, Seller shall sell and Buyer shall purchase the Products, including the Net Output delivered at the Point of Delivery. For and in consideration of Buyer's agreement to purchase from Seller the Net Output on the terms and conditions set forth herein, Seller transfers to Buyer, and Buyer accepts from Seller, all right, title, and interest that Seller may have in and to the associated RECs existing during the Term.

4.1.1 In addition, during the period between the Effective Date and the Commercial Operation Date, Seller shall sell and make available to Buyer, and Buyer shall purchase and receive, the Test Energy.

4.2 <u>No Sales to Third Parties</u>. During the Term, Seller shall not sell any portion of the Products to any Person other than Buyer; *provided, however*, that this restriction shall not apply to the extent (i) such sales are permitted pursuant to Section 11.3 or Section 14.3, or (ii) Buyer has committed an Event of Default that is not attributable to a DSP Failure.

4.3 <u>Title and Risk of Loss</u>. Seller shall deliver the Products and the Test Energy free and clear of all liens, security interest, claims and encumbrances. Title to and risk of loss of the Net Output and the Test Energy shall transfer from Seller to Buyer upon its delivery to Buyer at the Point of Delivery. Seller shall be deemed to be in exclusive control of, and responsible for, any damage or injury caused by the Net Output and the Test Energy up to the Point of Delivery. Buyer shall be deemed to be in exclusive control of, and responsible for, any damages or injury caused by the Net Output and the Test Energy at and from the Point of Delivery. All proceeds received by Buyer from the resale by Buyer of any portion of Products and the Test Energy shall belong exclusively to Buyer.

4.4 <u>Curtailment</u>. The rights and obligations of the Parties with respect to curtailments of energy from the Facility are as follows:

4.4.1 Involuntary Curtailments. Seller shall not be obligated to sell and make available energy from the Facility (or the other Products), and Buyer shall not be obligated to purchase, receive or pay for energy from the Facility (or the other Products), that is not delivered to the Point of Delivery due to any of the following (singularly, "Involuntary Curtailment" and collectively, "**Involuntary Curtailments**"): (a) to the extent resulting from an Emergency, or (b) during times and to the extent that an event of Force Majeure prevents either Party from delivering or receiving energy from the Facility, provided that, with respect to (b), a Party shall not be relieved of its obligations pursuant to this Section 4.4.1 if the underlying cause for the occurrence of the Force Majeure was within the reasonable control of such Party or any Person retained by such Person. For each Involuntary Curtailment, the total lost production caused by such Involuntary Curtailment, expressed in MWh ("Deemed Generated Energy from Curtailments"), shall be the Net Output that would have been produced by the Facility and delivered by Seller to Buyer at the Point of Delivery pursuant to this Agreement but that was not produced by the Facility and delivered to the Point of Delivery pursuant to this Agreement due to such Involuntary Curtailment, as reasonably determined by Seller using Expected Annual Production for the applicable Contract Year, Facility availability information, relevant weather conditions, solar insolation at the Facility and other pertinent data for the relevant period.

4.4.2 Buyer Initiated Curtailment. If at any time there is a Buyer Initiated Curtailment, then Seller shall not be obligated to sell and make available energy from the Facility (or the other Products). For each Buyer Initiated Curtailment, the total lost production caused by the Buyer Initiated Curtailment, expressed in MWh ("**Deemed Delivered Output**"), shall be the Net Output that would have been produced by the Facility and delivered by Seller to Buyer at the Point of Delivery pursuant to this Agreement but that was not produced by the Facility and delivered to the Point of Delivery pursuant to this Agreement due to such Buyer Initiated Curtailment, as reasonably determined by Seller using Expected Annual Production for the applicable Contract Year, Facility availability information, relevant weather conditions, solar insolation at the Facility and other pertinent data for the relevant period. Buyer shall compensate Seller for each MWh of Deemed Delivered Output at the Contract Price. Additionally, Buyer shall be responsible for any charges or penalties assessed by Transmission Owner or any Electric System Authority against either Party as a result of any Buyer Initiated Curtailment.

4.5 Environmental Attributes, Capacity Rights and Ancillary Services.

4.5.1 Purchase and Sale of Environmental Attributes. For and in consideration of Buyer's agreement to purchase from Seller the Net Output on the terms and conditions set forth herein, Seller will deliver all Environmental Attributes associated with the Net Output to Buyer. The Parties will make such filings, execute such periodic documentation and take all actions as are reasonably required to deliver documentation of the transfer of Environmental Attributes to Buyer through the Acceptable REC Registry. If the Acceptable REC Registry is unavailable, then Seller will deliver the Environmental Attributes to Buyer in an Environmental Attribute attestation to Buyer or other legal form to be agreed to by the Parties. Title and risk of loss of the Environmental Attributes shall transfer from Seller to Buyer at the time the Environmental Attributes are transferred to Buyer's Acceptable REC Registry account or at the time the Environmental Attributes are otherwise transferred to Buyer. 4.5.2 Registration. Seller shall cooperate in any registration reasonably requested by Buyer of the Facility in any renewable portfolio standard or other equivalent program in which Buyer may wish Seller to register or maintain registration of the Facility, to the extent that Buyer pays the costs and expenses associated therewith; provided, that in no event shall Seller be required to make any modifications or upgrades to the Facility or the Interconnection Facilities or to modify the Facility's operations in any manner in order to provide Environmental Attributes or register or comply with any renewable portfolio standard or other equivalent program.

4.5.3 Purchase and Sale of Capacity Rights. For and in consideration of Buyer's agreement to purchase from Seller the Net Output on the terms and conditions set forth herein, Seller transfers to Buyer, and Buyer accepts from Seller, any right, title, and interest that Seller may have in and to the Capacity Rights, if any, existing during the Term and associated with the Net Output to Buyer. It is acknowledged and agreed by the Parties that Seller shall not be obligated to incur any incremental costs or expenses (a) in developing, constructing or operating the Facility in order for Seller to provide Capacity Rights to Buyer or (b) to accredit the capacity of the Facility. In no event shall Seller be required to make any modifications or upgrades to the Facility or the Interconnection Facilities or to modify the Facility's operations in any manner in order to provide Capacity Rights.

4.5.4 Representation Regarding Ownership of Capacity Rights. Seller represents that it has not sold, and, subject to Section 4.2, covenants that during the Term it will not sell or attempt to sell to any Person, the Capacity Rights, if any. Subject to Section 4.2, during the Term, Seller shall not report to any Person that the Capacity Rights, if any, belong to anyone other than Buyer. Buyer may at its own risk and expense report to any Person that the Capacity Rights exclusively belong to it.

4.5.5 Purchase and Sale of Ancillary Services. For and in consideration of Buyer's agreement to purchase from Seller the Net Output on the terms and conditions set forth herein, Seller transfers to Buyer, and Buyer accepts from Seller, any right, title, and interest that Seller may have in and to the Ancillary Services, if any, existing during the Term and associated with the Net Output to Buyer. It is acknowledged and agreed by the Parties that Seller shall not be obligated to incur any incremental costs or expenses in developing, constructing or operating the Facility in order for Seller to provide Ancillary Services to Buyer. In no event shall Seller be required to make any modifications or upgrades to the Facility or the Interconnection Facilities or to modify the Facility's operations in any manner in order to provide Ancillary Services.

4.5.6 Representation Regarding Ownership of Ancillary Services. Seller represents that it has not sold, and, subject to Section 4.2, covenants that during the Term it will not sell or attempt to sell to any Person, the Ancillary Services, if any. Subject to Section 4.2, during the Term, Seller shall not report to any Person that the Ancillary Services, if any, belong to anyone other than Buyer. Buyer may at its own risk and expense report to any Person that the Ancillary Services exclusively belong to it.

4.5.7 Further Assurances. At Buyer's request, and, as between Buyer and Seller, at Buyer's sole cost and expense, the Parties shall execute such documents and instruments as may be reasonably required to effect recognition and transfer of the Capacity Rights and Ancillary Services, if any, to Buyer.

4.6 <u>Other Products</u>. This Agreement does not create for Buyer any rights or interests in any product generated by the Facility, other than Net Output, associated Environmental Attributes, associated Capacity Rights and associated Ancillary Services. Seller shall retain all rights, title and interest in and to (i) any and all existing or future products that are produced by or in any manner attributable to the Facility (other than Net Output, Environmental Attributes, Capacity Rights and Ancillary Services, in each case as generated by the Facility during the Term), (ii) Tax Credits or any local, state or federal cash grants, depreciation deductions or other tax credits providing a tax benefit to Seller or any other Person based on ownership of, or energy production from, any portion of the Facility, and (iii) cash grants, depreciation deductions and other tax benefits arising from ownership or operation of the Facility.

SECTION 5 CONTRACT PRICE; COSTS

5.1 <u>Contract Price</u>. Commencing on the Commercial Operation Date and continuing through the Term, Buyer shall pay the Contract Price for all deliveries to Buyer of the Products. The Contract Price includes the consideration to be paid by Buyer to Seller for the Products, and Seller shall not be entitled to any compensation over and above the Contract Price for the Products, except as set forth in Section 4.4.2.

5.1.1 For the period prior to the Commercial Operation Date, Buyer shall pay seventy-five percent (75%) of the Contract Price for all deliveries to Buyer of Test Energy.

5.2 Payment Obligation. Buyer's payment obligations associated with a DSP Failure under this Agreement (including any fees, charges, penalties or interest required hereunder unless expressly provided to the contrary) are limited to the amounts actually paid to Buyer by each Solar Participant pursuant to its respective PPC. Buyer hereby conveys, assigns, pledges and grants to Seller a security interest in all amounts actually paid by each Solar Participant pursuant to its PPC. The payment obligation of each Solar Participant pursuant to its PPC shall be supported by its full faith and credit taxing power and, in the event a Solar Participant becomes a Defaulting Solar Participant, Buyer hereby commits to Seller that it shall promptly pursue each of its remedies available to it pursuant to (x) the Defaulting Solar Participant's PPC, including, if necessary enforcement of the Defaulting Solar Participant's taxing power and (y) the non-defaulting Solar Participant. Buyer further represents and warrants that it has delivered to Seller substantially true and correct copies of each PPC and Buyer covenants that it shall not amend any PPC if to do so would have an adverse effect on Seller's rights or expected revenues under this Agreement.

5.3 <u>Costs and Charges</u>. Seller shall be responsible for all Liabilities, costs or charges imposed in connection with the delivery of Net Output up to the Point of Delivery, including line losses and any operation and maintenance charges imposed by the Transmission Owner for the Interconnection Facilities. Buyer shall be responsible for all Liabilities, costs or charges imposed in connection with the scheduling and transmission of Net Output from and beyond the Point of Delivery, except as otherwise provided in Section 6.1.2.

5.4 <u>Taxes</u>. Seller shall pay or cause to be paid when due, or reimburse Buyer for, all sales, use, severance, excise, ad valorem, and any other similar taxes imposed or levied by any

Governmental Authority on the generation, sale or delivery of the Net Output up to the Point of Delivery, regardless of whether such taxes are imposed on Buyer or Seller under Applicable Law. Buyer shall pay or cause to be paid when due, or reimburse Seller for, all such taxes levied at and beyond the Point of Delivery upon and after the purchase by Buyer of the Net Output, regardless of whether such taxes are imposed on Buyer or Seller under Applicable Law. Seller shall be responsible for all federal, state and local taxes of whatever kind relating to the construction, ownership, leasing, operation or maintenance of the Facility, the Premises, or any components or appurtenances thereof, and all of Seller's income taxes, including those based upon the sale of the Products. If a Party is required to remit or pay taxes that are the other Party's responsibility hereunder, the Party required to pay such taxes shall provide prompt written notice thereof to the Party responsible for such taxes, together with appropriate supporting documentation. The paying Party shall remit such taxes to the relevant Governmental Authority and the responsible Party shall reimburse the paying Party for such taxes. Such reimbursement shall be made by the responsible Party on or before ninety (90) days after (a) such notice and supporting documentation are received, or (b) such taxes are actually paid and proper documentation thereof is furnished, whichever is later. Any Party entitled to an exemption from any such taxes or charges shall furnish the other Party any necessary documentation related thereto. The Parties specifically reserve the right to protest to the appropriate state or political subdivision the amount or validity of any such taxes, whether or not any such action must be filed in the name of Seller, Buyer or both. At the responsible Party's expense, the paying Party shall reasonably cooperate with the responsible Party with any such action.

5.5 <u>Costs of Ownership and Operation</u>. Without limiting the generality of any other provision hereof, Seller shall be solely responsible for paying when due (a) all costs of owning and operating the Facility in compliance with existing and future Applicable Law and the terms and conditions hereof, and (b) all taxes and charges (however characterized) now existing or hereinafter imposed on or with respect to the Facility, its operation, or on or with respect to emissions or other environmental impacts of the Facility, including any such tax or charge (however characterized) to the extent payable by a generator of such energy or environmental attributes.

5.6 <u>Rates Not Subject to Review</u>. Neither Party shall petition FERC pursuant to the provisions of sections 205 or 206 of the Federal Power Act (16 U.S.C. § 792 *et seq.*) or any other Governmental Authority to amend this Agreement, or support a petition by any other Person seeking to amend this Agreement, absent the agreement in writing of the other Party. Further, absent the agreement in writing by both Parties, the standard of review for changes hereto proposed by a Party, a non-party or the FERC acting *sua sponte* shall be the "public interest" application of the "just and reasonable" standard of review set forth in *United Gas Pipe Line Co. v. Mobile Gas Corp.*, 350 U.S. 332 (1956) and *Fed. Power Comm'n v. Sierra Pacific Power Co.*, 350 U.S. 348 (1956) and clarified by *Morgan Stanley Capital Group*, *Inc. v. Pub. Util. Dist. No. 1 of Snohomish County*, 554 U.S. 527 (2008) and *NRG Power Mktg.*, *LLC v. Maine Pub. Utils. Comm'n*, 558 U.S. 165 (2009).

SECTION 6 OPERATION AND CONTROL

6.1 <u>Standard of Facility Operation</u>.

6.1.1 General; Operating and Forecast Procedures. Seller shall develop, build, operate, maintain and repair the Facility in all material respects in accordance with: (a) the applicable and mandatory standards, criteria and guidelines of the Transmission Owner and any Electric System Authority; (b) the Permits; (c) the Generation Interconnection Agreement; (d) all Applicable Laws; (e) the requirements of this Agreement; and (f) Good Industry Practices. Buyer and Seller shall cooperate to develop prior to the Commercial Operation Date (i) operating procedures for the scheduling, delivery and receipt of Test Energy and the Products as described hereunder, and (ii) procedures for providing non-binding forecasting of expected Net Output, in each case, in accordance with the requirements of this Agreement.

6.1.2 Fines and Penalties.

(a) Seller shall pay when due all fines, penalties, or legal costs incurred by Seller or for which Seller is legally responsible for noncompliance by Seller, its agents, employees, contractors or subcontractors, with respect to any provision hereof, any agreement, commitment, obligation or liability incurred by Seller in connection with this Agreement or the Facility or any Applicable Law, except where such fines, penalties or legal costs are being contested in good faith by Seller, its agents, employees, contractors or subcontractors through appropriate proceedings.

(b) If fines, penalties, or legal costs are assessed against or incurred by Buyer on account of any action by any Governmental Authority or the Transmission Owner due to noncompliance by Seller with any Applicable Law, the Generation Interconnection Agreement or the provisions hereof, or if the performance of Seller or Buyer is delayed or stopped by order of any Governmental Authority or the Transmission Owner due to Seller's noncompliance with any Applicable Law, or the Generation Interconnection Agreement, Seller shall indemnify and hold harmless Buyer against any and all losses, liabilities, damages and claims suffered or incurred by Buyer as a result.

(c) Without limiting the generality of anything in this Section 6.1.2, Seller shall reimburse Buyer for all fees, damages, fines, penalties or legal costs imposed on Buyer by any Governmental Authority or other Person or paid to other utilities for violations to the extent caused by Seller, and likewise Buyer shall reimburse Seller for all fees, damages, fines, penalties or legal costs imposed on Seller by any Governmental Authority or other Person or paid to other utilities for violations to the extent caused by Buyer.

6.2 <u>Interconnection</u>. Seller shall be responsible for the costs and expenses associated with interconnection of the Facility at the Point of Delivery, including Interconnection Facilities

defined herein, the costs of any Network Upgrades and the costs of any upgrades for affected systems.

6.3 <u>Coordination with the Transmission Owner</u>. As between the Parties, Seller shall be responsible for the coordination and synchronization of the Facility and the Interconnection Facilities with the Transmission Owner. In the event there are unanticipated changes in FERC or any Electric System Authority rules related to the coordination and synchronization of the Facility and the Interconnection Facilities with the Transmission Owner sufficiently significant to change the benefits, risks and burdens held by the Parties under this Agreement, the Parties shall meet in good faith to adjust the terms of this Agreement to provide for the Parties the originally intended allocation of benefits, risks and burdens.

6.4 <u>Outages</u>.

6.4.1 Planned Outages. Commencing with the second (2nd) Contract Year, Seller shall provide Buyer with an annual forecast of Planned Outages for each Contract Year at least one (1) month, but no more than three (3) months, before the first day of that Contract Year. Buyer shall have a period of fifteen (15) calendar days following Seller's delivery of such annual forecast to review and provide Seller with any requested changes in writing. Seller shall promptly update such annual forecast, or otherwise change it, only to the extent that Seller is reasonably required to change it in order to comply with Good Industry Practices, or the terms of any of Seller's or its Affiliates' financing documents. Seller shall not schedule any Planned Outages in the daylight hours during the period starting May 15th and ending October 15th of each Contract Year, inclusive, that reduce the energy generation capability of the Facility, unless (a) such outage is required to maintain equipment warranties, (c) such outage is required in order to comply with Good Industry Practices or the Generation Interconnection Agreement or (d) the Parties agree otherwise in writing.

6.4.2 Maintenance Outages. If Seller reasonably determines that it is necessary to schedule a Maintenance Outage, Seller shall notify Buyer of the proposed Maintenance Outage as soon as practicable but in any event at least five (5) days before the outage begins (or such shorter period to which Buyer may reasonably consent, acting reasonably). Upon such notice, the Parties shall plan the Maintenance Outage to mutually accommodate the reasonable requirements of Seller and the service obligations of Buyer. Notice of a proposed Maintenance Outage shall include the expected start date and time of the outage, the decrease in available capacity of the Facility, and the expected completion date and time of the outage. Buyer shall promptly respond to such notice and may request reasonable modifications in the schedule for the outage. Seller shall use all reasonable efforts to comply with any request to modify the schedule for a Maintenance Outage, provided that such change has no substantial impact on Seller and is consistent with Good Industry Practices. Seller shall notify Buyer of any subsequent changes in the available capacity of the Facility as a result of such Maintenance Outage or any changes in the Maintenance Outage completion date and time. As soon as practicable, any notifications regarding Maintenance Outages given orally shall be confirmed in writing. Seller shall take all reasonable measures and exercise its reasonable efforts consistent with Good Industry Practices to minimize the frequency and duration of Maintenance Outages.

6.4.3 Forced Outages. Seller shall promptly provide to Buyer an oral report, via telephone to a number specified by Buyer, of any Forced Outage affecting more than ten percent (10%) of the Actual Capacity of the Facility. This report shall include the amount of the Actual Capacity of the Facility that will not be available because of the Forced Outage and the expected return date of such generation capacity. Seller shall promptly update the report as necessary to advise Buyer of changed circumstances. Seller shall take reasonable measures consistent with Good Industry Practices to avoid Forced Outages and to minimize their duration.

6.4.4 Notice of Deratings and Outages. Without limiting the foregoing, Seller will inform Buyer via telephone to a number specified by Buyer, of any major limitations, restrictions, deratings, or outages known to Seller affecting the Facility for the following day and will promptly update Seller's notice to the extent of any material changes in this information, with "major" defined as affecting more than ten percent (10%) of the Actual Capacity of the Facility.

6.5 <u>Buyer Transmission Services</u>. Beginning no later than [March 31], 2023 and continuing throughout the Term, Buyer shall be responsible for arranging and paying for all transmission service required to effectuate the receipt of Test Energy and Net Output at the Point of Delivery. As between Buyer and Seller, Buyer shall bear all responsibility, liability, costs, fees, penalties and any other expenses associated with any failures, errors or omissions solely due to Buyer's performance of such obligations, including the failure to timely perform such obligations in accordance with this Agreement or the requirements of any Electric System Authority. Buyer shall indemnify, hold harmless and reimburse Seller for any liability, costs, fees, penalties and any other expenses assessed against or incurred by Seller that are Buyer's responsibility pursuant to the preceding sentence.

6.6 Increase in Actual Capacity After the Final Completion Date. Seller shall provide Buyer with written notice if Seller or any Affiliate of Seller elects to build an additional solar project in the geographic vicinity of the Facility (such notice a "**ROFR Notice**"), which ROFR Notice shall include information regarding the additional solar project (including but not limited to anticipated capacity and pricing and timing of construction). Buyer may, but shall not be required to, elect to increase its Contract Amount by an amount of MWac up to the maximum capacity of such additional solar project by providing written notice to Seller within [sixty (60)] days of actual receipt of the ROFR Notice. If Buyer elects not to participate in the additional solar project (or otherwise fails to timely respond to the ROFR Notice), then Seller may proceed with construction and selling the resulting output to a third party; provided that Seller shall be required to submit a new ROFR Notice to Buyer in accordance with this Section 6.6 for Buyer's consideration if the price agreed upon with such third party is less than the price offered to Buyer pursuant to the declined ROFR Notice.

6.7 <u>Meteorological Data</u>. Seller shall install sufficient meteorological stations at the Facility to provide the capability of measuring and recording representative irradiance levels and other pertinent meteorological conditions, which meteorological data may be used to calculate Deemed Generated Energy from Curtailments and Deemed Delivered Output.

6.8 <u>Forecasting</u>. Seller shall prepare and provide Buyer with the Facility's forecasted Energy production and the Net Output. These non-binding forecasts of production, described in Subsections 6.8.1 through 6.8.4, will be determined and prepared with the intent of being as accurate as possible. Seller shall update a forecast any time information becomes available indicating a material change in the forecast relative to the most previously provided forecast.

6.8.1 Year-Ahead Forecasts. Seller shall, by December 1 of each year during the Term, provide Buyer with a forecast of each month's average-day Net Output from the Facility, by hour, for the following calendar year. This forecast shall include an expected range of uncertainty based on historical operating experience. Seller shall update the forecast for each month at least five (5) Business Days before the first Business Day of such month.

6.8.2 Week-Ahead Forecasts. By 1700 EPT on the Thursday preceding the immediately upcoming week of delivery, Seller shall provide Buyer with a daily forecast of deliveries for the upcoming week (Saturday through Friday).

6.8.3 Day-Ahead Forecasts. By 0700 EPT on the calendar day immediately preceding the day of delivery, Seller shall provide Buyer with an hourly forecast of deliveries for each hour of the next seven (7) days. In the event that Seller has any information to believe that the production from the Facility on any day will be materially lower or higher than what would otherwise be expected based on the forecasts provided, then Seller will inform Buyer of such circumstance by 0700 EPT on the preceding Business Day.

6.8.4 Intra-day Forecasts. In the event that Seller has any information to believe that the production from the Facility during any hour will be materially lower or higher than what would otherwise be expected based on the forecasts provided, then Seller will inform Buyer of such circumstance as soon as practical.

6.8.5 Communication. Seller shall communicate forecasts in a form, template, substance, and manner as requested by Buyer (e.g., Excel template), which form, template, substance, and manner may be reasonably modified by Buyer from time to time. Requested forecast data may include but is not limited to, location, forecast timestamp, site capacity, a flag for actual or forecasted data, available site capacity, energy, and reason for any capacity reduction for each hour of the next seven days.

6.9 Operational Records and Reports.

6.9.1 Monthly Reports. Within fifteen (15) days after the end of each calendar month during the Term from and after the Commercial Operation Date, Seller shall provide to Buyer a monthly operational report, in electronic format, in a form and substance reasonably acceptable to Buyer containing the following data:

(a) A summary of the energy production of the Facility during such month and the Net Output; and

(b) A summary of any other significant events related to the operation of the Facility for such month.

6.9.2 Other Information to be Provided to Buyer. Seller shall provide to Buyer such other information respecting the condition or operations of Seller or the Facility as Buyer may, from time to time, reasonably request.

6.9.3 Information to Governmental Authorities. Seller shall, as soon as practicable, upon written request from Buyer, assist Buyer by providing Facility data and reports, to the extent available, that are required by Buyer to be provided to any Governmental Authority. In the event that Buyer obtains confidential technical or proprietary information related to the Facility and belonging to a third party, and Buyer is required to provide such information to a Governmental Authority or otherwise required to provide such information under Applicable Law, Seller may, at its expense, seek a protective order or other appropriate remedy to prevent such information from being disclosed to the public, all as further described in Section 24.4, below.

6.10 <u>Production Guarantee</u>.

6.10.1 Guaranteed Production. Seller guarantees that during each Performance Measurement Period during the Term, the Net Output for such Performance Measurement Period and the associated RECs will be no less than eighty-five percent (85%) of the Expected Annual Production for each Contract Year during such Performance Measurement Period (the "**Minimum Production Guarantee**"). In determining the Net Output in any Performance Measurement Period for purposes of compliance with the Minimum Production Guarantee, Deemed Delivered Output and Deemed Generated Energy from Curtailments shall be credited to the calculation of such Net Output.

6.10.2 Liquidated Damages for Failure to Meet Guaranteed Production. If Seller fails to achieve the Minimum Production Guarantee in any Performance Measurement Period, then Seller shall pay Buyer, as liquidated damages and not as a penalty, an amount equal to Buyer's Cost to Cover. Each Party agrees and acknowledges that the damages Buyer would incur due to any failure to achieve the Minimum Production Guarantee would be difficult to predict with certainty and therefore the liquidated damages set forth in this Section 6.10.2 are a fair and reasonable calculation of such damages. Such liquidated damages shall be payable in lieu of actual damages and shall be Buyer's sole and exclusive remedy and Seller's sole and exclusive liability for any failure by Seller to achieve the Minimum Production Guarantee; provided, however, notwithstanding the foregoing, Buyer shall nevertheless have the right to terminate this Agreement as a result of an Event of Default by Seller under Section 11.1.2(f) and seek those related rights and remedies upon such termination.

6.10.3 Annual Output Guarantee Report. On or before the sixtieth (60th) Day following the end of each Performance Measurement Period, Buyer shall deliver to Seller a report (and supporting data) detailing whether Seller achieved the Minimum Production Guarantee for the most recently completed Performance Measurement Period, and, if applicable, calculations of the liquidated damages set forth in Section 6.10.2 that Seller owes for such Performance Measurement Period. Within ten (10) Business Days of providing such invoice, Seller shall pay to Buyer, by wire transfer of immediately available funds to an account specified in writing by Buyer or by any other means agreed to by the Parties in writing from time to time, the amount set forth as due in such invoice or as otherwise determined pursuant to the dispute resolution provisions set forth in Section 23.

6.11 <u>Access Rights</u>. Upon reasonable prior notice, during normal working hours, and subject to the safety requirements of Seller and Applicable Laws relating to workplace health and safety, Seller shall provide Buyer and its authorized agents, employees and inspectors with

reasonable access to the Facility: (a) for the purpose of reading or testing metering equipment, (b) for purposes of implementing Section 10.6, (c) as necessary to witness any acceptance tests, and (d) for other reasonable purposes at the reasonable request of Buyer. Buyer shall indemnify and release Seller against and from any and all Liabilities resulting from actions or omissions by the Buyer Indemnitees in connection with their access to the Premises or the Facility, except to the extent that such damages are caused by the intentional or grossly negligent act or omission of any Seller Indemnitee.

6.12 <u>Signage</u>. If Buyer determines it would be useful for public relations purposes to have a sign at the site identifying the Solar Participants, Seller will provide such signage after review and approval by Buyer.

SECTION 7 GENERATOR STATUS

7.1 <u>Authority to Make Sales</u>. From and after the Commercial Operation Date (or, if earlier, the date Seller commences the delivery of the Test Energy), Seller shall maintain its authority to sell Net Output hereunder.

SECTION 8 SELLER'S SECURITY AND CREDIT SUPPORT

8.1 <u>Seller Credit Support</u>. Seller shall provide Buyer with and maintain the Seller Credit Support. The Seller Credit Support shall be available to Buyer as security for Seller's payment obligations under this Agreement. The Seller Credit Support shall be maintained at the expense of Seller and shall be, as elected by Seller, in its sole discretion, in the form of Letter of Credit, Cash Deposit, or Guaranty or a combination thereof. Seller may change the form or forms of Seller Credit Support at any time and from time to time upon reasonable prior notice to Buyer. Seller shall be required to replenish the Seller Credit Support if and as it is depleted.

8.1.1 One or More Letters of Credit. If all or a portion of the Seller Credit Support is a Letter of Credit, such Letter of Credit must be issued for a minimum term of three hundred sixty (360) days. Seller shall cause the renewal or extension of such Letter of Credit for additional consecutive terms of three hundred sixty (360) days or more no later than thirty (30) days prior to each expiration date of such Letter of Credit. If such Letter of Credit is not renewed or extended as required herein or replaced by alternate instruments of Seller Credit Support in accordance with this Section 8.1, Buyer shall have the right to draw immediately upon the Letter of Credit and to place the amounts so drawn in an interest bearing escrow account in accordance with Section 8.1.2, until and unless Seller provides Buyer with a substitute Letter of Credit or other form of Seller Credit Support meeting the requirements of this Section 8.1.

8.1.2 One or More Cash Deposits. Seller Credit Support provided in the form of a Cash Deposit shall include a requirement for immediate notice to Buyer and Seller in the event that the Cash Deposit amounts held as Seller Credit Support do not at any time meet the required level for the Cash Deposit. Cash Deposit funds held in the deposit account may be deposited in a money-market fund, short-term treasury obligations, investment-grade commercial paper and other liquid investment-grade investments with maturities of three (3) months or less as may be specified by Seller, with all investment income thereon to be taxable to, and to accrue for the benefit of, Seller. At any such times as the balance in the deposit account exceeds the amount of Seller's obligation to provide the Cash Deposit hereunder, Buyer shall remit to Seller on demand any excess in the deposit account above the required level of the Cash Deposit.

8.1.3 One or more Guaranties. Seller may provide one or more Guaranties as Seller Credit Support. If the Guarantor with respect to any Guaranty provided as Seller Credit Support ceases to qualify as a Guarantor, then Seller shall be required to either (a) provide a replacement Guaranty from a qualified Guarantor, or (b) replace the Guaranty provided by the unqualified Guarantor with one or more alternate Seller Credit Support instruments meeting the criteria set forth in this Section 8.1, in each case, no later than ten (10) Business Days after such Guarantor ceases to be a qualified Guarantor.

8.1.4 Credit Support Release. Promptly following the date upon which Seller provides the Subsequent Credit Support (to the extent that the Construction Credit Support is not part of the Subsequent Credit Support) or upon earlier termination of this Agreement and payment of any outstanding Delay Damages, Buydown Liquidated Damages, and any other amounts due to Buyer, Buyer shall release the balance of the Construction Credit Support (including any accumulated interest, if applicable) to Seller. Promptly following the end of the Term or upon earlier termination of this Agreement, and payment of any outstanding amounts due to Buyer, Buyer shall release the balance of the Seller Credit Support (including any accumulated interest, if applicable) to Seller.

8.2 <u>Not an Exclusive Remedy</u>. The security contemplated by this Section 8 shall not be Buyer's exclusive remedy for the Seller's failure to perform its obligations in accordance with this Agreement.

SECTION 9 METERING AND COMMUNICATION

9.1 <u>Installation of Metering Equipment</u>. Seller shall be responsible for ensuring that metering equipment is designed, furnished, installed, owned, inspected, tested, calibrated, maintained and replaced as provided in the Generation Interconnection Agreement.

9.2 <u>Metering</u>. Metering shall be performed at the location and in the manner specified in the Generation Interconnection Agreement and as necessary to perform Seller's obligations hereunder.

9.2.1 Revenue Meters. Upon Buyer's reasonable request and no less than once per calendar year, Seller shall inspect and test the Revenue Meter. Buyer may have a representative present during any metering inspection or test. If any of the inspections or tests, whether requested by Buyer or Seller, disclose a variance exceeding one-half of one percent (0.5%), proper correction, based upon the found variance, shall be made of previous readings for the actual period during which the metering equipment rendered inaccurate measurements if that period can be ascertained. If the actual period cannot be ascertained, the proper correction shall be made to the measurements taken during the time the metering equipment was in service since last tested, but not exceeding the shorter of (a) the last one-half of the period from the last test to the test that

showed the metering equipment to be in error or (b) three (3) months preceding the test that showed the metering equipment to be in error, in the amount the metering equipment shall have been shown to be in error by such test. The Parties shall use production data from Seller's computer monitoring system for the Facility to determine the amount of such inaccuracy. Any correction in billings or payments resulting from a correction in the meter records shall be made in the next monthly billing or payment rendered; provided, however, that, no corrections shall be made with respect to any payments that were due more than two (2) years prior to the date of the test or inspection that showed the metering equipment to be in error. Such correction, when made, shall constitute full adjustment of any claim between Seller and Buyer arising out of such inaccuracy of metering equipment. Nothing in this Agreement shall give rise to Buyer having any obligations to Seller, or any other Person, pursuant to or under the Generation Interconnection Agreement.

9.2.2 Communication. Besides the communication equipment required in the Interconnection Agreement between the Seller and its transmission interconnection service provider, the Seller will provide a MEAG Power meter station to include communication equipment to allow MEAG Power to read the meters (primary and backup) in real time from MEAG Power's Supervisory Control and Data Acquisition (SCADA) system and MEAG Power's MV90 metering system in Atlanta, GA. All communication, SCADA and telemetry equipment will be provided by the Seller and will be designed to meet MEAG Power's equipment and data communication requirements which include, but are not limited to: primary revenue quality meter (with Power Quality capabilities), back-up revenue quality meter, two (2) private, redundant communication channels with related technology on both ends in order to deliver data from the facility to MEAG Power's SCADA and MV90 systems in Atlanta, GA, including redundant routers and network equipment designed to communicate real time information to MEAG Power.

SECTION 10 BILLINGS, COMPUTATIONS AND PAYMENTS

10.1 <u>Monthly Invoices</u>. No later than the seventh (7th) Business Day after the end of each calendar month, Seller shall deliver to Buyer an invoice detailing Seller's computation of the payment for the Net Output delivered to Buyer and any Deemed Delivered Output during the previous month. Together with the invoice, Seller shall provide detailed computations showing the Net Output that was delivered during each hour and calculations by hour for Deemed Delivered Output. All settlement statements will reflect hours in Central Prevailing Time. If any invoice is found to be inaccurate, a corrected invoice shall be issued. On or before the last day of the month during which the invoice was received, Buyer shall pay Seller the undisputed portion of the invoiced amount for Net Output and for Deemed Delivered Output.

10.2 <u>Offsets</u>. Either Party may offset any undisputed payment due hereunder against amounts owing from the other Party to the offsetting Party pursuant hereto; provided that any offset with respect to an amount attributable to a Solar Participant shall be limited by and subject to the principles in Section 5.2. Offsets shall be documented in a manner mutually acceptable to the Parties, with records retained as otherwise required by this Agreement. A Party's exercise of recoupment and set off rights shall not limit the other remedies available to such Party.

10.3 <u>Interest on Late Payments</u>. Any amounts that are not paid when due hereunder (including any amounts withheld by a Party which are later determined to have been improperly

withheld) shall bear interest at the Contract Interest Rate from the date due until paid, which rate shall not exceed the maximum permissible under Georgia law.

10.4 <u>Disputed Amounts</u>. Notwithstanding anything to the contrary herein, if either Party, in good faith, disputes any amount due pursuant to an invoice rendered under this Section 10, such Party shall notify the other Party of the specific basis for the dispute and may, if not previously paid, withhold payment of the amount disputed. Upon resolution of the dispute, any required payment or refund shall be paid within thirty (30) days after such resolution, along with interest at the Contract Interest Rate from the date of the overpayment or underpayment to the date of receipt of the reconciling payment or refund.

10.5 <u>Recordkeeping</u>. Each Party shall keep and maintain all records as may be necessary or useful in performing or verifying the accuracy of all relevant data, estimates or statements of charges submitted hereunder until the later of (a) a period of at least two (2) years after the date an invoice was received by a Party, or (b) if there is a dispute relating to an invoice, the date that is at least two (2) years after the date on which such dispute is resolved. Each Party agrees to take such reasonable action as may be requested by the other Party to comply with the record retention requirements imposed by a Governmental Authority or that are otherwise necessary to preserve the tax benefits of this transaction.

10.6 Required Records and Audit Rights. Seller shall maintain, or cause to be maintained, accurate and up-to-date books and records of all matters relating to operations, production, billings and other terms of this Agreement in sufficient detail and format in order that all provisions of this Agreement can be readily and adequately verified. All accounting records shall be maintained in accordance with GAAP consistently applied. All required records shall be available in electronic format. Seller shall also maintain or cause to be maintained, accurate and up-to-date records as may be required by law, Governmental Authorities, and other regulators. Buyer shall have the right, at its sole expense, upon reasonable notice, and during normal business hours, to examine, audit and copy: (i) the records of Seller to the extent reasonably necessary to verify the accuracy of any invoice, charge or computation made hereunder; (ii) operating procedures and manuals; (iii) equipment and operating records, and (iv) data and records required to be maintained pursuant this Agreement; provided however that audits shall not be conducted more frequently than once each Contract Year and shall be limited in scope to the fiscal year of the audit as well as the immediately two preceding fiscal years. Seller will support remote audits as requested through transmission of requested items in an electronic format and participation in conference calls and webinars. If, as a result of such examination, any amount is determined to be due by one Party to the other Party, the appropriate Party shall promptly pay the amount of the deficiency so owed, together with interest at the Contract Interest Rate from the date of the overpayment or underpayment to the date of receipt of the reconciling payment.

10.7 <u>Payment Due Dates</u>.

Except as otherwise specifically provided herein, any payment due from one Party to the other Party pursuant to this Agreement shall be due thirty (30) days after the date of the invoice that the former Party receives from the latter Party for the amount of such payment.

10.8 <u>Rounding</u>. In administering and interpreting this Agreement, all amounts expressed in MWh shall be rounded to the nearest thousandth of a MWh, all amounts expressed in MW shall be rounded to the nearest thousandth of a MW, all rates per MW and rates per MWh amounts expressed in dollars shall be rounded to the nearest hundredth of a dollar, and all total amounts expressed in dollars shall be rounded to the nearest dollar.

SECTION 11 DEFAULTS AND REMEDIES

11.1 <u>Defaults</u>. The following events are defaults (each a "default" before the passing of applicable notice and cure periods, and an "Event of Default" thereafter) hereunder:

11.1.1 Defaults by Either Party.

(a) A Party breaches a representation or warranty made by it herein if such breach materially affects the performance of such Party's obligations under this Agreement and the breach is not cured within thirty (30) days after the nondefaulting Party gives the defaulting Party a written notice of the default. A breach may only be "cured" by the defaulting party remedying the breach and reimbursing the non-defaulting party for any cost, expense, or liability incurred as a result of such breach (including, but not limited to, any cost or expense incurred investigating and pursuing such breach).

(b) A Party fails to perform any material obligation hereunder for which an exclusive remedy is not provided hereunder and which is not addressed in any other Event of Default described in this Section 11.1, if the failure is not cured within thirty (30) days after the non-defaulting Party gives the defaulting Party written notice of the default; *provided*, *however*, that, upon written notice from the defaulting Party, this thirty (30) day period shall be extended by an additional ninety (90) days if (i) the failure cannot reasonably be cured within the original thirty (30) day period despite diligent efforts, (ii) the default is capable of being cured within the additional ninety (90) day period, and (iii) the defaulting Party commences the cure within the original thirty (30) day period and is at all times thereafter diligently proceeding to cure the failure.

11.1.2 Defaults by Seller.

(a) Seller (i) makes a general assignment for the benefit of its creditors; (ii) files a petition or otherwise commences, authorizes or acquiesces in the commencement of a proceeding or cause of action under any bankruptcy or similar law for the protection of creditors, or has such a petition filed against it and such petition is not withdrawn or dismissed within sixty (60) days after such filing; (iii) becomes insolvent; or (iv) is generally unable to pay its debts when due.

(b) Seller fails to make an undisputed payment when due hereunder (including any liquidated damages or other payments upon default due hereunder) if the failure is not cured within fifteen (15) days after the Buyer gives Seller a written notice of the failure.

(c) Seller sells or transfer the Products to any other Person other than Buyer, except as otherwise permitted under this Agreement and such activity is not cured within ten (10) Business Days after Buyer gives Seller a written notice thereof.

(d) Seller fails to post or maintain Seller Credit Support as required in Section 8 if the failure is not cured within ten (10) Business Days after Buyer gives Seller a written notice of such failure.

(e) Except as provided in Section 20.3, Seller transfers all or substantially all of the assets of the Facility without the prior written consent of Buyer.

(f) If during any two (2) consecutive Performance Measurement Periods, the aggregate quantity of Net Output is less than sixty-five percent (65%) of the aggregate sum of the Expected Annual Production for such period, provided that in determining the Net Output in any Performance Measurement Period, Deemed Delivered Output and Deemed Generated Energy from Curtailments shall be credited to the calculation of such Net Output.

(g) Seller fails to deliver the Environmental Attributes and the Test Energy free and clear of all liens, security interests, claims and encumbrances if the failure is not cured within ten (10) Business Days after Buyer gives Seller a written notice of such failure.

(h) Seller fails to issue notice to its contractor (which contractor shall have been previously disclosed to Buyer) under the engineering, procurement and construction agreement or similar contract entered into by and between Seller and the contractor and relating to the construction of the Facility, authorizing and directing the commencement of construction of the Facility by the Notice to Proceed Date.

11.1.3 Default by Buyer.

(a) Buyer fails to make an undisputed payment when due hereunder and:

(1) If such failure to pay is not attributable to a Defaulting Solar Participant, such failure is not cured within fifteen (15) days after the Seller gives the Buyer a written notice of the failure to pay; or

(2) If such failure to pay is attributable to a Defaulting Solar Participant (a "**DSP Failure**"), at the time of such failure Buyer fails to promptly pursue each of its available remedies under (i) the Defaulting Solar Participant's PPC, including, if necessary, enforcement of the Defaulting Solar Participant's taxing power and (ii) the non-defaulting Solar Participants' PPC obligations to make payments arising from the nonpayment of a Defaulting Solar Participant. (b) Buyer (i) makes a general assignment for the benefit of its creditors; (ii) files a petition or otherwise commences, authorizes or acquiesces in the commencement of a proceeding or cause of action under any bankruptcy or similar law for the protection of creditors, or has such a petition filed against it and such petition is not withdrawn or dismissed within sixty (60) days after such filing; (iii) becomes insolvent; or (iv) is generally unable to pay its debts when due.

11.2 <u>Termination and Remedies</u>.

11.2.1 Buyer's Remedies. Upon the occurrence of an Event of Default by Seller (excluding any Event of Default arising out of any event or circumstance for which an exclusive remedy is expressly provided under this Agreement), the Buyer shall be entitled to all remedies available under this Agreement or at law or in equity, and may terminate this Agreement by notice to Seller designating the date of termination (provided such notice must be delivered to the Seller to the notice addresses of the Seller set forth in Section 22.1 no less than fifteen (15) Business Days before such termination date). Such notice shall be sent by registered overnight delivery service or by certified or registered mail, return receipt requested and shall state prominently therein that the notice is a notice of termination of this Agreement. In the event of a termination hereof, and without prejudice to any other remedies available to Buyer under this Agreement or at law or in equity (except as to remedies which are liquidated as hereafter provided in Section 11.2.1(c), below):

(a) Seller shall pay to Buyer all amounts due to Buyer hereunder for all periods prior to termination, subject to offset by either Party against amounts due from the other Party for periods prior to such termination.

(b) The amounts due pursuant to Section 11.2.1(a) shall be paid within sixty (60) days after receipt of invoice for such charges, or following a determination pursuant to the dispute resolution provisions set forth in Section 23 if applicable, and shall bear interest thereon at the Contract Interest Rate from the date of termination until the date paid. The foregoing does not extend the due date of, or provide an interest holiday for, any payments otherwise due hereunder.

(c) The Buyer shall be entitled to receive a Termination Payment as liquidated damages for all amounts owing under this Agreement for the balance of the Term. The Termination Payment shall be calculated by the Buyer within a reasonable period after termination of this Agreement. Amounts owed pursuant to this Section 11.2.1(c) shall be due within sixty (60) days after the Buyer gives the Seller notice of the amount due or following a determination pursuant to the dispute resolution provisions set forth in Section 23 if applicable. Subject to Section 11.3, the Buyer shall under no circumstances be required to account for or otherwise credit or pay the Seller for economic benefits accruing to the Buyer as a result of the Seller's default, except to the extent such benefits are contemplated in the definition of Termination Payment.

(d) In the exercise of any remedy under this Section 11.2.1, Buyer shall be entitled to recover from Seller, to the extent permitted by Georgia law, all

reasonable attorneys' fees and expenses incurred by Buyer in connection with the exercise of such remedies.

11.2.2 Seller's Remedies.

(a) Upon the occurrence of an Event of Default by Buyer under Section 11.1.1, Section 11.1.3(a)(1) or Section 11.1.3(b), Seller shall be entitled to all remedies available under this Agreement or at law or in equity, and may terminate this Agreement by notice to Buyer designating the date of termination (provided such notice must be delivered to Buyer to the notice addresses of the Buyer set forth in Section 22.1 no less than fifteen (15) Business Days before such termination date). Such notice shall be sent by registered overnight delivery service or by certified or registered mail, return receipt requested and shall state prominently therein that the notice is a notice of termination of this Agreement. In the event of a termination hereof, and without prejudice to any other remedies available to Seller under this Agreement or at law or in equity (except as to remedies which are liquidated as hereafter provided in Section 11.2.2(a)(3), below):

(1) Buyer shall pay to Seller all amounts due to Seller hereunder for all periods prior to termination, subject to offset by either Party against amounts due from the other Party for periods prior to such termination.

(2) The amounts due pursuant to Section 11.2.2(a)(1) shall be paid within sixty (60) days after receipt of invoice for such charges, or following a determination pursuant to the dispute resolution provisions set forth in Section 23 if applicable, and shall bear interest thereon at the Contract Interest Rate from the date of termination until the date paid. The foregoing does not extend the due date of, or provide an interest holiday for, any payments otherwise due hereunder.

(3) The Seller shall be entitled to receive a Termination Payment as liquidated damages for all amounts owing under this Agreement for the balance of the Term. The Termination Payment shall be calculated by the Seller within a reasonable period after termination of this Agreement. Amounts owed pursuant to this Section 11.2.2(a)(3) shall be due within sixty (60) days after the Seller gives the Buyer notice of the amount due or following a determination pursuant to the dispute resolution provisions set forth in Section 23 if applicable. Subject to Section 11.3, the Seller shall under no circumstances be required to account for or otherwise credit or pay the Buyer for economic benefits accruing to the Seller as a result of the Buyer's default, except to the extent such benefits are contemplated in the definition of Termination Payment.

(4) In the exercise of any remedy under this Section 11.2.2(a), Seller shall be entitled to recover from Buyer, to the extent permitted by Georgia law, all reasonable attorneys' fees and expenses incurred by Seller in connection with the exercise of such remedies.

Upon the occurrence of an Event of Default that is attributable to a (b)DSP Failure, if Buyer is not, at any time, using its reasonable best efforts to pursue its available remedies against (i) the Defaulting Solar Participant pursuant to the terms of the Defaulting Solar Participant's PPC and (ii) the non-defaulting Solar Participants pursuant to the terms of their respective PPCs, then Seller may bring any suit, action or proceeding in law or in equity, including mandamus, injunction and action for specific performance as may be necessary or appropriate to cause Buyer to enforce such remedies (including but not limited to enforcement of the Defaulting Solar Participant's taxing power) until such DSP Failure is cured, and to the extent permitted by Georgia law, Seller may recover from Buyer all reasonable attorneys' fees and expenses incurred by Seller in connection with the exercise of Seller's remedy under this Section 11.2.2(b). Buyer shall periodically apprise Seller of steps taken by Buyer to pursue its remedies pursuant to clauses (i) and (ii) of the immediately preceding sentence. If any amount that is attributable to a DSP Failure remains unpaid to Seller 12 months following the month of the initial Event of Default attributable to the DSP Failure, then Seller shall have the right to terminate this Agreement with respect to the portion of the Contract Amount attributable to the Defaulting Solar Participant.

11.2.3 Notwithstanding any termination or expiration of this Agreement, this Agreement will remain in effect with respect to, and to the extent necessary to facilitate the settlement of, all liabilities and obligations arising hereunder before the effective date of such termination or expiration, and as necessary to facilitate any Termination Payment.

11.2.4 The Parties agree to pursue the remedies available hereunder in such manner as reasonably required to permit the continued performance of the remaining obligations.

11.3 <u>Duty to Mitigate</u>. Each Party agrees that it has a duty to mitigate damages and covenants that it will use commercially reasonable efforts to minimize any damages it may incur as a result of the other Party's performance or non-performance hereof.

11.4 <u>Security</u>. Buyer may, in addition to pursuing any and all other remedies available at law or in equity, proceed against any Seller Credit Support held by Buyer in whatever form to reduce any undisputed amounts that Seller owes Buyer arising from a Seller Event of Default.

11.5 <u>Cumulative Remedies</u>. Except with respect to events or circumstances for which an exclusive remedy is provided herein, the rights and remedies provided to each Party hereunder are cumulative and not exclusive of any rights or remedies of such Party.

SECTION 12 INDEMNIFICATION AND LIABILITY

12.1 <u>Indemnities</u>.

12.1.1 Indemnity by Seller. To the extent permitted by Applicable Law, Seller shall release, indemnify and hold harmless Buyer, its Affiliates and members, and each of its and their respective directors, officers, employees, agents, and representatives (collectively, the "**Buyer Indemnitees**") against and from any and all losses, fines, penalties, claims, demands, damages,

liabilities, actions or suits of any nature whatsoever (including reasonable legal costs and attorneys' fees, both at trial and on appeal, whether or not suit is brought) in each case as are claimed by third parties (collectively, "**Liabilities**") resulting from, or arising out of, or in any way connected with, Seller's breach of the performance of its obligations hereunder, gross negligence, willful misconduct, fraud, or violation of Applicable Law, for or on account of personal injury, illness, death or property damage, excepting only to the extent such Liabilities are caused by the gross negligence or willful misconduct of any Buyer Indemnitee.

12.1.2 Indemnity by Buyer. To the extent permitted by Applicable Law, Buyer shall release, indemnify and hold harmless Seller, its Affiliates, and each of its and their respective directors, officers, employees, agents, and representatives (collectively, the **"Seller Indemnitees**") against and from any and all Liabilities resulting from, or arising out of, or in any way connected with, Buyer's breach of the performance of its obligations hereunder, gross negligence, willful misconduct, fraud, or violation of Applicable Law, for or on account of personal injury, illness, death or property damage, excepting only to the extent such Liabilities are caused by the gross negligence or willful misconduct of any Seller Indemnitee.

12.1.3 Defense. Promptly after receipt by a Party of any claim or notice of the commencement of any action, administrative or legal proceeding, or investigation as to which the indemnity provided for in this Section 12 may apply, the indemnified Party shall notify the indemnifying Party in writing of such fact. The indemnifying Party shall assume the defense thereof with counsel designated by such indemnifying Party and satisfactory to the indemnified Party, *provided, however*, that if the defendants in any such action include both the indemnified Party and the indemnifying Party and the indemnified Party shall have reasonably concluded that there may be legal defenses available to it which are different from or additional to, or inconsistent with, those available to the indemnifying Party, the indemnified Party shall have the right to select and be represented by separate counsel, at the indemnifying Party's expense.

12.1.4 Failure to Defend. If the indemnifying Party fails to assume the defense of a claim meriting indemnification, the indemnified Party may at the sole expense of the indemnifying Party, contest, settle, or pay such claim; *provided however*, that settlement or full payment of any such claim may be made only following consent of the indemnifying Party or, absent such consent, written opinion of the indemnified Party's counsel that such claim is meritorious or warrants settlement.

12.1.5 Third Parties. Except as provided in this Section 12.1 and as is consistent to preserve the Financing Parties' rights in any Financing Party Consent or Estoppel, nothing herein shall be construed to create any duty to, any standard of care with reference to, or any liability to any Person not a Party hereto. No undertaking by one Party to the other under any provision hereof shall affect the status of Buyer or Seller as an independent entity.

12.2 <u>Consequential Damages</u>. EXCEPT TO THE EXTENT SUCH DAMAGES ARE INCLUDED IN ANY LIQUIDATED DAMAGES OR TERMINATION PAYMENT PAYABLE HEREUNDER OR INDEMNIFICATION FOR THIRD PARTY DAMAGES HEREUNDER OR FOR LIABILITY ARISING FROM FRAUD, NEITHER PARTY SHALL BE LIABLE TO THE OTHER PARTY FOR SPECIAL, PUNITIVE, INDIRECT, EXEMPLARY OR CONSEQUENTIAL DAMAGES, WHETHER SUCH DAMAGES ARE ALLOWED OR PROVIDED BY CONTRACT, TORT (INCLUDING NEGLIGENCE), STRICT LIABILITY, STATUTE OR OTHERWISE.

SECTION 13 INSURANCE

13.1 Evidence of Insurance. On or before the Notice to Proceed Date and continuing through achievement of Commercial Operation, Seller shall acquire, or cause to be acquired, Builder's Risk insurance coverage as required by Exhibit 2. On or before the beginning of each Contract Year, Seller shall acquire, or cause to be acquired, all other insurance required by Exhibit 2. In connection with all insurance coverage required to be carried by Seller hereunder, Seller shall provide Buyer, upon Buyer's prior written request, with certificates of insurance evidencing that insurance coverages for the Facility are in compliance with the applicable specifications for such insurance coverage set forth in Exhibit 2. Such insurance policies shall (a) provide Buyer with additional insured status (except worker's compensation/employer's liability); and (b) provide a waiver of any rights of subrogation against Buyer, its Affiliates and their officers, trustees, directors, agents, subcontractors, and employees. If the insurer does not agree to provide Buyer thirty (30) days' prior written notice of non-renewal, cancellation of corresponding policies (except that such notice shall be ten (10) days for non-payment of premiums), Seller shall be obligated to provide such notice to Buyer. All insurance policies shall be written with insurers rated A-VIII or better by A.M. Best Company. All insurance policies shall be written on an occurrence or claims-made basis. All insurance policies, except the Umbrella/Excess Liability policy, shall contain an endorsement or otherwise provide that Seller's policy shall be primary in all instances regardless of like coverage, if any, carried by Buyer. Seller's liability under this Agreement is not limited to the amount of insurance coverage required herein. The Umbrella/Excess Liability policy shall be non-contributory.

13.2 <u>Term and Modification of Insurance</u>. If any insurance required to be maintained by Seller hereunder ceases to be reasonably available and commercially feasible in the commercial insurance market, Seller shall provide written notice to Buyer, accompanied by a certificate from an independent insurance advisor, certifying that such insurance is not reasonably available and commercially feasible in the commercial insurance market for electric generating plants of similar type, geographic location and design. Upon receipt of such notice, Seller shall use commercially reasonable efforts to obtain other insurance that would provide comparable protection against the risk to be insured and Buyer shall not unreasonably withhold its consent to such modification.

SECTION 14 FORCE MAJEURE

14.1 <u>Definition of Force Majeure</u>. "Force Majeure" or "an event of Force Majeure" means an event that (a) is not within the reasonable control of the Party affected by the event, (b) is not the result of such Party's negligence or failure to act, and (c) could not be overcome, avoided or mitigated by the affected Party's use of due diligence under the circumstances. Force Majeure includes, but is not restricted to, events of the following types (but only to the extent that such an event, in consideration of the circumstances, satisfies the tests set forth in the preceding sentence): acts of God; natural disasters; fire; severe weather; storms; lightning; tsunami; peril of the sea; war (declared or undeclared); military or guerilla action; banditry; terrorist activity or a threat of

terrorist activity which, under the circumstances, would be considered a precursor to actual terrorist activity; economic sanction or embargo; pandemic, epidemic or quarantine; civil disturbance; sabotage; action, inaction or restraint by court order or public or Governmental Authority. Notwithstanding the foregoing, none of the following constitute Force Majeure: (i) Seller's ability to sell, or Buyer's ability to purchase, the Products at a more advantageous price than is provided hereunder; (ii)(a) the unavailability, variability or lack of photovoltaic rays or solar insolation, or (b) Facility equipment failures, in each case, except to the extent caused by an independent event of Force Majeure; or (iii) economic hardship, including lack of money.

14.2 <u>Suspension of Performance</u>. If either Party is rendered wholly or in part unable to perform its obligations hereunder because of an event of Force Majeure, both Parties shall be excused from the performance affected by the event of Force Majeure (other than the obligation to pay amounts due hereunder), provided that:

14.2.1 the Party affected by the Force Majeure shall give the other Party prompt written notice, without intentional delay, describing the particulars of the event;

14.2.2 the suspension of performance shall be of no greater scope and of no longer duration than is required to remedy the effect of the Force Majeure; and

14.2.3 the affected Party shall use reasonable and diligent efforts to remedy its inability to perform.

14.3 <u>Force Majeure Does Not Affect Obligations Already Incurred</u>. No obligations, including payment obligations, of either Party that arose before the event of Force Majeure or that arise after the cessation of the event of Force Majeure shall be excused by the event of Force Majeure. In the event of an event of Force Majeure with respect to Buyer, Seller may sell all or a portion of the Products to any other Person during the continuation of such event of Force Majeure.

14.4 <u>Strikes</u>. Notwithstanding any other provision hereof, neither Party shall be required to settle any strike, walkout, lockout or other labor dispute on terms which, in the sole judgment of the Party involved in the dispute, are contrary to the Party's best interests.

14.5 <u>Right to Terminate</u>. If a Force Majeure event prevents a Party from performing its material obligations hereunder for a period exceeding three hundred sixty-five (365) consecutive days, then either Party may terminate this Agreement by giving ten (10) days' prior written notice to the other Party, provided that if the Party claiming to be affected by the Force Majeure event is diligently taking reasonable efforts to remedy the effects of the Force Majeure event, then the Party claiming the Force Majeure event shall have such additional time (not to exceed six (6) additional months) as reasonably necessary to remedy its inability to perform. Upon such termination, each Party shall pay to the other all amounts due the other hereunder for all periods prior to the date of termination, but neither Party will have any liability to the other Party (a) as a result of such termination; provided, however, that this Agreement will remain in effect with respect to, and to the extent necessary to, facilitate the settlement of, all liabilities and obligations arising hereunder before the effective date of such termination.

14.6 <u>Change in Law</u>. In the event of an enactment, adoption, promulgation, amendment, modification, repeal or change in interpretation by a Governmental Authority of any Applicable Law after the Effective Date that impairs or prevents either Party's performance of its material obligations under this Agreement, materially increases a Party's costs to perform its material obligations under this Agreement, or materially alters the allocation of risks between the Parties (a "**Change in Law**"), the Parties shall use good faith efforts to negotiate and agree on modifications to this Agreement in a manner that preserves the respective economic benefits and risk allocation of the Parties under this Agreement as of the Effective Date; *provided, however*, that neither Party shall be obligated to accept or agree to any modifications to this Agreement, nor shall any Party be bound to take any action or perform any obligation unless such action or obligation has been mutually agreed upon by the Parties.

SECTION 15 SEVERAL OBLIGATIONS

Nothing contained herein shall be construed to create an association, trust, partnership or joint venture or to impose a trust, partnership or fiduciary duty, obligation or liability on or between the Parties or any Solar Participant.

SECTION 16 CHOICE OF LAW

This Agreement shall be interpreted and enforced in accordance with the laws of the State of Georgia, excluding any choice of law rules that may direct the application of the laws of another jurisdiction.

SECTION 17 PARTIAL INVALIDITY

The Parties do not intend to violate any Applicable Law governing the subject matter hereof. If any of the terms hereof are finally held or determined to be invalid, illegal, unenforceable, or void as being contrary to any Applicable Law or public policy, all other terms hereof shall remain in effect. The Parties shall use good faith efforts to amend this Agreement to reform or replace any terms determined to be invalid, illegal or void, such that the amended terms (a) comply with and are enforceable under Applicable Law, (b) give effect to the intent of the Parties in entering hereinto, and (c) preserve the balance of the economics and equities contemplated by this Agreement in all material respects.

SECTION 18 NON-WAIVER

No waiver of any provision hereof shall be effective unless the waiver is set forth in a writing that (a) expressly identifies the provision being waived, and (b) is executed by the Party waiving the provision. A Party's waiver of one or more failures by the other Party in the performance of any of the provisions hereof shall not be construed as a waiver of any other failure, whether of a like kind or different nature.

SECTION 19 AUTHORIZATIONS

During the Term, Seller shall maintain all Permits required, as applicable, for the construction, operation, and ownership of the Facility and for the sale and delivery of the Net Output, except Permits with respect to which the failure to maintain would not materially adversely affect Seller's ability to perform its obligations hereunder.

SECTION 20 SUCCESSORS AND ASSIGNS

20.1 <u>Restriction on Assignments</u>. Except as expressly provided in this Section 20, neither Party may assign this Agreement or any of its rights or obligations hereunder, without the prior written consent of the other Party, which consent shall not be unreasonably withheld, conditioned or delayed. Any assignment in contravention of this Section 20 will be void.

20.2 <u>Binding Nature</u>. This Agreement shall be binding upon and inure to the benefit of the permitted successors and assigns of the Parties hereto.

Permitted Assignments by Seller. Seller may not assign this Agreement or any 20.3 portion thereof to any Person without the prior written consent of Buyer (such consent not to be unreasonably withheld, conditioned or delayed); provided, however, Seller may, without the consent of Buyer, assign this Agreement to a Financing Party for collateral security purposes in connection with any financing or refinancing of the Facility, and in connection therewith, Buyer agrees to (i) execute a reasonable Financing Party Consent or Estoppel in a form reasonably acceptable to Buyer should the Financing Party request such consent, (ii) reasonably cooperate in a timely manner with the requests of Seller or Financing Parties in conjunction with any financing involving the Facility, including any due diligence efforts of any such Financing Parties, and (iii) deliver reasonable and customary legal opinions, if required, in connection with any Financing Party Consent or Estoppel that is entered into with or for a Financing Party. If Seller seeks Buyer's consent to transfer or assign this Agreement to any party succeeding to all or substantially all of the assets or generating assets of Seller, such proposed transferee (together with the proposed transferee's Affiliates) must at the time of assignment, (x) either (A) collectively own, manage, or operate solar or wind energy electricity generating assets (not including the Facility) with a nameplate capacity of not less than 200 MW in the aggregate or (B) have engaged (or will have engaged in connection with such assignment) a Person to manage the construction of the Facility (if such transfer occurs before the Commercial Operation Date) or the operation of the Facility (if such transfer occurs after the Commercial Operation Date) that, together with its Affiliates, collectively owns, manages, or operates solar or wind energy electricity generating assets with a nameplate capacity of not less than 200 MW in the aggregate; (y) have an Acceptable Credit Rating or a credit rating at least equal to that of the assignor; and (z) have posted Credit Support in accordance with this Agreement. In connection with any permitted assignment by Seller and approved by Buyer pursuant to the immediately preceding sentence, Buyer will execute a consent and agreement to such assignment and a written full release of Seller from all of its duties and obligations under this Agreement in a form reasonably acceptable to the Parties.

Permitted Assignments by Buyer. Notwithstanding Section 20.1, Buyer may, upon 20.4 Seller's written consent (such consent not to be unreasonably withheld, conditioned or delayed) assign a portion of its rights and obligations under this Agreement to a Person (including a joint action agency) with whom Buyer may be working for the supply of prepaid gas or electricity. To the extent this Agreement is novatable pursuant to the supply agreements between such Person and the prepaid gas or electricity seller, and to the extent Buyer wishes to novate this agreement, Buyer will deliver a written request for such assignment, which request will include a proposed assignment agreement. Subject to the consent required by the first sentence of this Section 20.4, Seller agrees to (i) comply with the prepaid-gas or electricity seller's reasonable requests for knowyour-customer and similar account opening information and documentation with respect to Seller, including but not limited to information related to forecasted generation, credit rating, and compliance with anti-money laundering rules, the Dodd-Frank Act, the Commodity Exchange Act, the Patriot Act and similar rules, regulations, requirements and corresponding policies; and (ii) promptly execute such assignment agreement and implement such assignment as contemplated thereby, subject only to the countersignature of the appropriate parties involved with Buyer's supply of prepaid gas or electricity.

SECTION 21 ENTIRE AGREEMENT

This Agreement supersedes all prior agreements, proposals, representations, negotiations, discussions or letters, whether oral or in writing, regarding the subject matter hereof. No modification hereof shall be effective unless it is in writing and executed by both Parties.

SECTION 22 NOTICES

22.1 <u>Addresses and Delivery Methods</u>. All notices, requests, statements or payments shall be made to the addresses set out below. Notices required to be in writing shall be delivered by letter, facsimile, electronic mail, or other written documentary form. Notice by hand delivery shall be deemed to have been given when received or hand delivered. Notice by overnight mail or courier shall be deemed to have been given on the date and time evidenced by the delivery receipt.

To Seller:	PINEVIEW SOLAR LLC Attn: General Counsel 685 S. Arthur Ave., Suite 1B Louisville, CO 80027		
with a copy to	PINEVIEW SOLAR LLC Attn: Alexander Zhou 685 S. Arthur Ave., Suite 1B Louisville, CO 80027		

To Buyer:	MEAG Power Attn: Senior VP and Chief Operating Officer 1470 Riveredge Parkway, NW Atlanta, GA 30328-4640
with a copy (which shall not constitute notice) to:	MEAG Power Attn: Senior VP and General Counsel 1470 Riveredge Parkway, NW Atlanta, GA 30328-4640

22.2 <u>Changes of Address</u>. The Parties may change any of the persons to whom notices are addressed, or their addresses, by providing written notice in accordance with this Section 22.

22.3 <u>Notices to Financing Parties</u>. The requirements concerning notice by Buyer to Financing Parties will be set forth in the Financing Party Consent or Estoppel, if any.

SECTION 23 DISPUTE RESOLUTION

23.1 <u>Negotiations</u>. The Parties shall attempt in good faith to resolve all disputes arising out of, related to or in connection with this Agreement promptly in accordance with this Section 23.1. Any Party may give the other Party written notice of any dispute not resolved in the normal course of business. Executives of both Parties at levels at least one level above the personnel who have previously been involved in the dispute shall meet at a mutually acceptable time and place within fifteen (15) days after delivery of such notice, and thereafter as often as they reasonably deem necessary, to exchange relevant information and to attempt to resolve the dispute. If the matter has not been resolved within thirty (30) days after the referral of the dispute to such senior executives, or if no meeting of such senior executives has taken place within fifteen (15) days after such referral, either Party may initiate litigation as provided hereinafter.

23.2 <u>Choice of Forum</u>. Any legal action or proceeding arising out of this Agreement or the actions of the Parties leading up to this Agreement must be brought in the Superior Court of Fulton County in the State of Georgia. By execution and delivery hereof, each Party (a) accepts the exclusive jurisdiction of such court and waives any objection that it may now or hereafter have to the exercise of personal jurisdiction by such court over each Party for the purpose of any proceeding related to this Agreement, (b) irrevocably agrees to be bound by any final judgment (after any and all appeals) of such court in any such proceeding, (c) irrevocably waives, to the fullest extent permitted by law, any objection that it may now or hereafter have to the laying of venue of any such suit, action or proceedings brought in such court (including any claim that any such suit, action or proceeding has been brought in an inconvenient forum) in connection herewith, (d) agrees that service of process in any such action may be effected by mailing a copy thereof by registered or certified mail, postage prepaid, to such Party at its address as set forth herein, and I agrees that nothing herein shall affect the right to effect service of process in any other manner permitted by law.

23.3 <u>Settlement Discussions</u>. No statements of position or offers of settlement made in the course of the negotiation process described in Section 23.1 may be offered into evidence for any purpose in any litigation between the Parties, nor will any such statements or offers of settlement be used in any manner against either Party in any such litigation. Further, no such statements or offers of settlement shall constitute an admission or waiver of rights by either Party in connection with any such litigation. At the request of either Party, any such statements and offers of settlement, and all copies thereof, shall be promptly returned to the Party providing the same.

23.4 <u>Waiver of Jury Trial</u>. EACH PARTY KNOWINGLY, VOLUNTARILY, INTENTIONALLY AND IRREVOCABLY WAIVES THE RIGHT TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION BASED ON THIS AGREEMENT, OR ARISING OUT OF, UNDER OR IN CONNECTION WITH THIS AGREEMENT.

SECTION 24 CONFIDENTIALITY

24.1 Confidential Information; Press Releases. The Parties hereby agree to keep confidential, and shall not disclose (including by means of press release or other public announcement), the terms of this Agreement, any information relating to the business, strategy, policies, prospects, assets or plans of the other Party or any of the other Party's Affiliates (including, in the case of Buyer as the receiving Party, Facility capabilities and availability and other information contained in monthly and quarterly reports provided hereunder) and, to the extent marked in writing as confidential at the time of disclosure, all other information provided by the Parties to one another pursuant to this Agreement (collectively, "Confidential Information"). Confidential Information shall not include (i) information that is or becomes available to the public through no breach of this Agreement, (ii) information that was previously known by the receiving Party without any obligation to hold it in confidence, (iii) information that the receiving Party receives from a third party who may disclose that information without breach of law or agreement, (iv) information that the receiving Party develops independently without using the Confidential Information, and (v) information that the providing Party approves for release in writing. Buyer and Seller hereby agree that the Parties will coordinate and mutually agree upon the content and timing of press releases, public announcements, marketing materials, and branding materials concerning this Agreement or the subject matter of this Agreement. Notwithstanding the foregoing, the other Party's prior written approval shall not be required nor prevent the Parties from releasing information (a) which is required to be disclosed in order to obtain Permits and other approvals relating to the Facility or other information required to be provided to regulators having jurisdiction over a Party's business or that of its Affiliates, (b) as necessary to fulfill such Party's obligations under this Agreement, or (c) as otherwise required by Applicable Law.

24.2 <u>Treatment of Confidential Information</u>. Notwithstanding the foregoing, each Party may provide any Confidential Information: (i) to a Governmental Authority or any other Person (including contractors, consultants, accountants, financial advisors, agents, experts, legal counsel and other professional advisors to the Parties) as required for billing or otherwise to perform its obligations under or administer this Agreement; and (ii) in the case of Seller, to Financing Parties or potential Financing Parties, Affiliates and lessors, owners of and potential bidders and bidders

for, and potential purchasers and purchasers of, direct or indirect interests in the Facility or Seller and to any credit rating agency that has issued a Credit Rating for Seller or any of its Affiliates. Each Party shall cause its personnel and all Persons to whom it discloses the Confidential Information to treat it confidentially and to not disclose it to any other Person in any manner whatsoever, except as permitted hereunder. The obligation to provide confidential treatment to Confidential Information shall not be affected by the inadvertent disclosure of Confidential Information by either Party.

24.3 Disclosure Required by Law. Notwithstanding the foregoing, a receiving Party may use and disclose Confidential Information where required to do so in litigation, administrative, regulatory or other legal proceedings or otherwise by Applicable Law, but (to the extent reasonably possible under Applicable Law) only after providing written notice to the providing Party and affording the providing Party an opportunity to seek a protective order or other relief to prevent or limit disclosure of the Confidential Information. In such event, the receiving Party shall (to the extent permitted under Applicable Law) reasonably cooperate in connection with the providing Party's efforts to obtain such protective order or other relief. In no event shall a receiving Party be required to provide such a written notice to the providing Party when making a required disclosure in any litigation where the Parties are adverse litigants. Further, each Party shall use commercially reasonable efforts to maintain the confidentiality of the Confidential Information in any litigation or administrative or regulatory proceeding or in any other instance where disclosure is required by Applicable Law and there is an opportunity to maintain the confidentiality of the Confidential Information under the Applicable Law, and shall promptly notify the providing Party of any attempt by any Person to obtain the Confidential Information through legal process or otherwise under Applicable Law. In addition, to the extent either Party requires additional details or other information from the other Party in order to comply with Applicable Law and such additional details or other information is in such other Party's possession or reasonably available to such other Party, such other Party shall promptly provide such additional details or other information to the first Party upon request therefor from the first Party to such other Party.

24.4 Georgia Open Records Act and Georgia Open Meeting Act.

24.4.1 Notwithstanding any other provision of this Section 24 or this Agreement, the Parties recognize Buyer and the Solar Participants are governmental entities subject to certain statutory, legal obligations as a public body, including, without limitation, the Georgia Open Records Act (O.C.G.A. §50-18-70, *et seq.*) ("**GORA**") and Georgia Open Meeting Act (O.C.G.A. § 50-14-1, *et seq.*) ("**GOMA**"). The obligations and requirements of Buyer under GORA and GOMA shall be harmonized to the extent feasible with this section, provided that GORA and GOMA, as interpreted by Georgia courts, shall control. Upon receipt of a request for Confidential Information and prior to the release of any Confidential Information by Buyer pursuant to GORA or GOMA, Buyer will provide written notice to Seller of such request as set forth in Section 24.3, above, along with the information or records which Buyer has identified as responsive which may contain Confidential Information. If Seller should object to the production of any Confidential Information by Buyer under GORA or GOMA, Seller must timely notify Buyer of its objection. Buyer shall make an independent determination as to whether Confidential Information is required to be disclosed pursuant to GORA and GOMA. Seller may, if Seller disagrees with Buyer's determination, institute an action in its own name to prevent or otherwise limit such disclosure.

Each Party agrees to bear their own costs, including legal fees, in complying with this Section 24.4.

24.4.2 Upon Buyer's receipt of a Solar Participant's written notification (a "Written Notification"), in accordance with the terms of such Solar Participant's PPC, that such Solar Participant has received a request for a copy of this Agreement pursuant to GORA, Buyer shall use commercially reasonable efforts to promptly notify Seller of such GORA Request. Notwithstanding anything to the contrary in this Agreement, a Solar Participant's failure to provide the Written Notification and/or Buyer's failure to notify Seller upon receipt of a Written Notification (i) shall not constitute a default pursuant to Section 11.1, hereof and shall not give rise to any of the Seller remedies referenced in Section 11.2.2, hereof and (ii) Buyer shall have no recourse against the Solar Participant or Buyer.

24.5 <u>Survival</u>. The obligations of the Parties under this Section 24 will remain in full force and effect for three (3) years following the expiration or termination of this Agreement.

SECTION 25 COUNTERPARTS; ORIGINALS

This Agreement may be executed by facsimile or PDF (electronic copy), in one or more duplicate counterparts, and when executed and delivered (electronically or manually) by all the parties listed below, shall each be deemed an original and, taken together, shall constitute a single binding agreement.

[signature page follows]

IN WITNESS WHEREOF, the Parties have caused this Agreement to be executed in their respective names as of the date first above written.

SELLER:

PINEVIEW SOLAR LLC

By:		
Name:		
Title:		

By:		
Name:		
Title:		

BUYER:

MUNICIPAL ELECTRIC AUTHORITY OF GEORGIA

By:_______ Name:___James E. Fuller Title:__CEO and President
EXHIBIT A

DESCRIPTION OF FACILITY AND POINT OF DELIVERY

The Facility shall consist of a solar photovoltaic (PV) inverter-based generator array with a nominal net capacity of 80 MW_{AC} at the Point of Delivery, which Point of Delivery will be the 115 kV bus at MEAG Power's Pitts 230/115 kV Substation. The Facility site is located approximately at GPS coordinates Latitude 32.004384, Longitude –83.544910, in Wilcox County, Georgia.

EXHIBIT B

EXPECTED ANNUAL PRODUCTION for

Contract Amount from Facility

Contract Year	MWh (80 MWac)
1	196,149
2	195,168
3	194,192
4	193,221
5	192,255
6	191,294
7	190,338
8	189,386
9	188,439
10	187,497
11	186,559
12	185,626
13	184,698
14	183,775
15	182,856
16	181,942
17	181,032
18	180,127
19	179,226
20	178,330

EXHIBIT C

NERC EVENT TYPES

Event	Description of Outages
Туре	A O
U1	<u>Unplanned (Forced) Outage—Immediate</u> – An outage that requires immediate
	removal of a unit from service, another outage state or a Reserve Shutdown state.
	This type of outage results from immediate mechanical/electrical/hydraulic control
	systems trips and operator-initiated trips in response to unit alarms.
U2	Unplanned (Forced) Outage—Delayed – An outage that does not require
	immediate removal of a unit from the in-service state but requires removal within
	six (6) hours. This type of outage can only occur while the unit is in service.
U3	<u>Unplanned</u> (Forced) Outage—Postponed – An outage that can be postponed
	beyond six hours but requires that a unit be removed from the in-service state
	before the end of the next weekend. This type of outage can only occur while the
	unit is in service.
SF	<u>Startup Failure</u> – An outage that results from the inability to synchronize a unit
	within a specified startup time period following an outage or Reserve Shutdown.
	A startup period begins with the command to start and ends when the unit is
	synchronized. An SF begins when the problem preventing the unit from
	synchronizing occurs. The SF ends when the unit is synchronized or another SF
МО	Occurs.
MO	<u>Maintenance Outage</u> – An outage that can be deferred beyond the end of the next weekend, but requires that the unit be removed from service before the next
	planned outage. (Characteristically, a MO can occur any time during the year, has
	a flexible start date, may or may not have a predetermined duration and is usually
	much shorter than a PO.)
ME	Maintenance Outage Extension – An extension of a maintenance outage (MO)
	beyond its estimated completion date. This is typically used where the original
	scope of work requires more time to complete than originally scheduled. Do not
	use this where unexpected problems or delays render the unit out of service
	beyond the estimated end date of the MO.
РО	Planned Outage – An outage that is scheduled well in advance and is of a
	predetermined duration, lasts for several weeks and occurs only once or twice a
	year. (Boiler overhauls, turbine overhauls or inspections are typical planned
	outages.)
PE	Planned Outage Extension – An extension of a planned outage (PO) beyond its
	estimated completion date. This is typically used where the original scope of
	work requires more time to complete than originally scheduled. Do not use this
	where unexpected problems or delays render the unit out of service beyond the
	estimated end date of the PO.

EXHIBIT D

EXAMPLE CALCULATION OF BUYER'S COST TO COVER

Assume that during a Performance Measurement Period the Expected Annual Production is 175,000 MWh; however, during the Performance Measurement Period the actual Net Output was 125,000 MWh, representing only 71.43% of the Expected Annual Production.

Also assume that several Involuntary Curtailments were mandated by the Transmission Owner (or its transmission operator) within the Performance Measurement Period. The Seller determines that the Net Output that would have been produced by the Facility but that was not produced by the Facility due to such Involuntary Curtailments amounts to 350 MWh. This determination was based on the Expected Annual Production, Facility availability information, relevant weather conditions, and solar insolation at the Facility and other pertinent data for the relevant period.

Finally, assume that there were no Buyer Initiated Curtailments during the Performance Measurement Period, therefore, the Deemed Delivered Output is 0 MWh.

Because the actual Contract Amount of Net Output for such Performance Measurement Period and the associated RECs are less than Minimum Production Guarantee of 85%, Seller is to pay to Buyer liquidated damages in an amount equal to Buyer's Cost to Cover.

Buyer's Cost to Cover is calculated as follows:

(Replacement Cost \$/MWh - Contract Price \$/MWh) * ((85%* Expected Annual Production MWh) - (actual Net Output + Involuntary Curtailments MWh + Buyer Initiated Curtailments MWh))

Where,

Replacement Cost = The average of the market price that MEAG Power posts for its Participants for all hours ending 900 through hours ending 1700 in the Performance Measurement Period, plus MEAG's actual price paid for RECs to replace RECs not delivered by Seller, all on a \$/MWh basis. For this example, assume MEAG's average posted market price for the hours HE900 through HE1700 for the Performance Measurement Period is \$30.00/MWh, and MEAG's actual cost of replacement RECs for the Performance Measurement Period is \$2.00/MWh.

Then,

Buyer's Cost to Cover = (\$32.00 - \$25.91) * ((\$5% * 175,000.00) - (125,000 + 350 + 0))= (\$6.09) * (148,750 - 125,350)= \$142,506.00

EXHIBIT E

FORM OF LETTER OF CREDIT

[LETTERHEAD OF ISSUING BANK]

Irrevocable Standby Letter of Credit No.

Issued: November ____, 2020

Beneficiary: Municipal Electric Authority of Georgia 1470 Riveredge Pkwy Atlanta, GA 30328

Applicant: [NAME]

Initial expiration date at our counter (unless evergreen):

Final expiration date at our counter:

For overnight delivery:

Ladies and Gentlemen:

We, [**] ("Issuer") do hereby issue this Irrevocable Transferable Standby Letter of Credit No. by order of and for the account of [NAME] ("Account Party") and in favor of Municipal Electric Authority of Georgia ("Beneficiary" or "You"), in connection with that certain Power Purchase Agreement dated ______, 2021, by and between Account Party and You (as amended, modified and supplemented from time to time, the "Agreement"). The term "Beneficiary" includes any successor by operation of law of the named beneficiary including without limitation any liquidator, receiver or conservator. Capitalized terms used herein but not defined shall have the meaning given such terms in the Agreement.

This Letter of Credit is issued, presentable and payable and we guaranty to you that drafts under and in compliance with the terms of this Letter of Credit will be honored on presentation and surrender of certain documents pursuant to the terms of this Letter of Credit.

This Letter of Credit is available in one or more drafts and may be drawn hereunder, in part or in full, for the account of up to an aggregate amount not exceeding ______. This Letter of Credit is drawn against by presentation to us at our office located at [**, Attention: **], of a drawing

certificate: (i) signed by an officer or authorized agent of the Beneficiary; (ii) dated the date of presentation; and (iii) the following statement:

"The undersigned hereby certifies to [**] ("Issuer"), with reference to its Irrevocable Transferable Standby Letter of Credit No., dated , issued on behalf of [NAME] ("Account Party") and in favor of Municipal Electric Authority of Georgia ("Beneficiary") that (i) the Account Party failed to furnish a replacement or extension letter of credit (or other substitute credit support) thirty (30) days prior to the final expiration date of the Letter of Credit; (ii) the Account Party failed to furnish a replacement letter of credit (or other substitute credit support) within thirty (30) days after the Issuer notified us that it elected not to consider the letter of credit extended for one (1) year; (iii) the Issuer no longer qualifies as a Qualified Issuer; or (iv) an Event of Default with respect to the Account Party has occurred under the Agreement. Based upon the foregoing and in accordance with the terms of Section 8 of the Agreement, the Beneficiary hereby draws upon the Letter of Credit in an amount equal to \$ (United States Dollars)."

If presentation of any drawing certificate is made on a Business Day and such presentation is made on or before 10:00 a.m. Eastern Time, Issuer shall satisfy such drawing request on the same Business Day. If the drawing certificate is received after 10:00 a.m. Eastern Time, Issuer will satisfy such drawing request on the next Business Day. As used herein, "Business Day" means any day on which (a) commercial banks are not closed, or authorized or required to close, in New York City, New York and (b) with respect to certain drawing request, the bank to which funds are requested to be transferred hereunder as set forth in such drawing certificate is not closed, or authorized or required to close, and may receive such funds by wire transfer as requested hereunder.

It is a condition of this Letter of Credit that it will be automatically extended without amendment for one (1) year from the initial expiration date hereof, or any future expiration date subject to the final expiration date hereof, unless at least one hundred twenty (120) days prior to any expiration date we send you written notice at the above address by registered mail or overnight courier service that we elect not to consider this Letter of Credit extended for any such period.

However, in no event shall this Letter of Credit be extended beyond the final expiration date.

This Letter of Credit may be transferred in its entirety (but not in part) upon presentation to us of a transfer certificate signed by the Beneficiary in the form of Exhibit A accompanied by this Letter of Credit and any amendment(s), in which the Beneficiary irrevocably transfers to such transferee all of its rights hereunder, whereupon we agree to either issue a substitute letter of credit to such successor or endorse such transfer on the reverse of this Letter of Credit. Any transfer fees assessed by Issuer will be payable solely by Account Party, and the payment of any transfer fees will not be a condition to the validity or effectiveness of the transfer or this Letter of Credit.

This Letter of Credit is not transferable to any person or any entity with which U.S. persons are prohibited from doing business under applicable U.S. law or regulation.

Disbursements under the Letter of Credit shall be in accordance with the following terms and conditions:

1. The amount, which may be drawn by the Beneficiary under this Letter of Credit, may be reinstated by the amount of any drawings hereunder via amendment.

2. All commissions and charges will be borne by the Account Party.

3. This Letter of Credit shall be governed by the International Standby Practices Publication No. 590 of the International Chamber of Commerce (the "ISP"), except to the extent that terms hereof are inconsistent with the provisions of the ISP, in which case the terms of the Letter of Credit shall govern. This Letter of Credit shall be governed by the internal laws of the State of New York to the extent that the terms of the ISP are not applicable; provided that, in the event of any conflict between the ISP and such New York laws, the ISP shall control.

4. This Letter of Credit may not be amended, changed or modified without the express written consent of the Beneficiary and the Issuer.

5. The Beneficiary shall not be deemed to have waived any rights under this Letter of Credit, unless the Beneficiary or an authorized agent of the Beneficiary shall have signed a written waiver.

No such waiver, unless expressly so stated therein, shall be effective as to any transaction that occurs subsequent to the date of the waiver, nor as to any continuance of a breach after the waiver. Partial drawing permitted.

6. A failure to make any partial drawing at any time shall not impair or reduce the availability of this Letter of Credit in any subsequent period or our obligation to honor your subsequent demands for payment made in accordance with the terms of this Letter of Credit.

7. All payments hereunder shall be made free and clear of, and without deduction or set off for or on account of, any present or future taxes, duties, charges, fees, deductions or withholdings of any nature and by whomsoever imposed.

Exhibit A

(FORM TO BE ADDRESSED TO THE NOMINATED BANK BY THE BENEFICIARY OF A TRANSFERABLE CREDIT WHEN TRANSFERRING THE CREDIT IN ITS ENTIRETY INCLUDING ALL EXISTING AND FUTURE AMENDMENTS, IF ANY)

То:

Gentlemen:

Re: Letter of Credit No. Issued by: [**]

For value received, the undersigned beneficiary hereby irrevocably transfers to:

(Name of Transferee)

(Address)

all rights of the undersigned beneficiary to Draw under the above Letter of Credit In its entirety.

By this transfer, all rights of the undersigned beneficiary in such Letter of Credit are transferred to the transferee and the transferee shall have the sole rights as beneficiary thereof, including sole rights relating to any amendments whether increases or extensions or other amendments and whether now existing or hereafter made. All amendments are to be advised direct to the transferee without necessity of any consent of or notice to the undersigned beneficiary.

The advice of such Letter of Credit is returned herewith, and we ask you to endorse the transfer on the reverse thereof, and forward it direct to the transferee with your customary notice of transfer.

Very truly yours,

Signature of Beneficiary

SIGNATURE AUTHENTICATED & SIGNOR IS AUTHORIZED TO REQUEST SAID TRANSFER

(Bank)

(Authorized Signature)

EXHIBIT F

FORM OF SELLER GUARANTY

This Guaranty is executed and delivered as of this ____ Day of _____, 20___ by _____, a _____ [corporation] ("<u>Guarantor</u>"), in favor of [____] ("<u>Buyer</u>"), in connection with the performance by [____] ("<u>Seller</u>"), of a Power Purchase Agreement dated [___] between Seller and Buyer (the "<u>PPA</u>").

- RECITALS -

A. Seller is planning to develop, design, construct, own, and operate a solar power electric generation facility expected name plate capacity of approximately [____] MW to be located in [____] (the "Facility").

B. Seller and Buyer have entered into the PPA for the purchase and sale of electrical energy and associated Environmental Benefits (as defined in the PPA) from the Facility on the terms and conditions set forth therein.

[Seller is controlled by Guarantor.] Guarantor expects to derive substantial benefits from the performance of the PPA by Seller and Buyer. To induce Buyer to enter into the PPA and consummate the purchase and sale of electrical energy contemplated by the PPA, Guarantor has agreed to guarantee the obligations of Seller as provided in this Guaranty.

NOW, THEREFORE, in consideration of the foregoing, Guarantor agrees as follows:

- AGREEMENT -

1. <u>Guaranty</u>. Subject to the provisions of this Guaranty, Guarantor hereby absolutely, irrevocably, unconditionally, and fully guarantees to Buyer the due, prompt, and complete observance, performance, and discharge of all amounts payable by Seller under the PPA when the same shall become due and payable, whether on scheduled payment dates, upon demand, upon declaration of termination, upon acceleration or otherwise, and whether incurred before or after the date of delivery of this Guaranty (the "<u>Obligations</u>"). The obligations of Guarantor under this Guaranty are primary obligations for which Guarantor is the principal obligor. There are no conditions precedent to the enforcement of this Guaranty, except as expressly contained herein. This is a guaranty of payment, not of collection and shall apply regardless of whether recovery of all such Obligations may be discharged, or uncollectible in any bankruptcy, insolvency, reorganization, liquidation, receivership, or similar proceeding affecting Seller or its assets. Buyer shall not be required to institute, pursue, or exhaust any remedies against Seller before instituting suit, obtaining judgment, and executing thereon against Guarantor under this Guaranty.

2. <u>Maximum Liability</u>. Notwithstanding anything herein to the contrary, Guarantor's maximum liability under this Guaranty shall be limited to (\$US_____), plus costs of collection with respect to any valid claim(s) made by Buyer hereunder that are incurred in the enforcement or protection of the rights of Buyer.

3. <u>Guaranty Absolute</u>. The liability of Guarantor under this Guaranty shall be absolute, irrevocable and unconditional irrespective of:

(a) any defect or deficiency, or other unenforceable provision, in the PPA or any other documents executed in connection with the PPA;

(b) except as to applicable statutes of limitation, failure, omission, delay, waiver or refusal by Buyer to exercise, in whole or in part, any right or remedy held by Buyer with respect to the PPA, including but not limited to those rights set forth in Section 4 of this Guaranty; or

(c) any change in the existence, structure or ownership of Guarantor or Seller, or any bankruptcy, insolvency, reorganization, liquidation, receivership, or similar proceeding affecting Seller or its assets.

4. <u>Rights of Buyer</u>. Guarantor hereby grants to Buyer, in Buyer's discretion and without the need to notify or obtain any consent from Guarantor, and without termination, impairment, or any other effect upon Guarantor's duties hereunder, the power and authority from time to time:

(a) to renew, compromise, extend, accelerate, or otherwise change, substitute, supersede, or terminate the terms of performance of any of the Obligations, in each case in accordance with the PPA;

(b) to enter into any amendments, supplements or modifications to the PPA;

(c) to grant any indulgences, forbearances, and waivers, on one or more occasions, for any length of time, with respect to Seller's performance of any of the Obligations; and

(d) to accept collateral, further guaranties, and/or other security for the Obligations, and, if so accepted, then to impair, exhaust, exchange, enforce, waive, or release any such security.

5. <u>Performance</u>. If any of the Obligations are not performed according to the tenor thereof ("<u>Default</u>"), Guarantor shall within five (5) days of receipt of written demand by Buyer (a) perform or cause Seller to perform the Obligation in Default, and (b) pay, reimburse, and indemnify Buyer against any liabilities, damages, and related costs (including attorneys' fees) incurred by Buyer as a result thereof, all in such manner and at such times as Buyer may reasonably direct.

6. <u>Satisfaction</u>. Satisfaction by Guarantor of any duty hereunder incident to a particular Default or the occurrence of any other Default shall not discharge Guarantor except with respect to such Default that was satisfied, it being the intent of Guarantor that this Guaranty be continuing until such time as all of the Obligations have irrevocably been discharged in full, at which time this Guaranty shall automatically terminate. If at any time the performance of any Obligation by Seller or Guarantor is rescinded or voided under the federal Bankruptcy Code or otherwise, then Guarantor's duties hereunder shall continue and be deemed to have been

automatically reinstated, restored, and continued with respect to that Obligation, as though the performance of that Obligation had never occurred, regardless of whether this Guaranty otherwise had terminated or would have been terminated following or as a result of that performance.

7. Guarantor shall be subrogated to all rights of Buyer against Seller in respect of any amounts paid by Guarantor pursuant to this Guaranty, provided that Guarantor waives any rights it may acquire by way of subrogation under this Guaranty, by any payment made hereunder or otherwise (including any statutory rights of subrogation under Section 509 of the Bankruptcy Code, 11 U.S.C. §509, or otherwise), reimbursement, exoneration, contribution, indemnification, or any right to participate in any claim or remedy of Buyer against Seller or any collateral which Buyer now has or acquires, until all of the Obligations shall have been irrevocably paid to Buyer in full.

8. <u>Notice of Acceptance</u>. Guarantor waives and acknowledges notice of acceptance of this Guaranty by Buyer.

9. <u>Waivers by Guarantor</u>. Guarantor hereby waives and agrees not to assert or take advantage of:

(a) all set-offs, counterclaims, and, subject to Section 5 above, all presentments, demands for performance, notices of non-performance, protests, and notices of every kind that may be required by applicable laws;

(b) any right to require Buyer to proceed against Seller or any other person, or to require Buyer first to exhaust any remedies against Seller or any other person, before proceeding against Guarantor hereunder;

- (c) any defense based upon an election of remedies by Buyer;
- (d) any duty of Buyer to protect or not impair any security for the Obligations;
- (e) the benefit of any laws limiting the liability of a surety;

(f) any duty of Buyer to disclose to Guarantor any facts concerning Seller, the PPA or the Facility, or any other circumstances, that would or allegedly would increase the risk to Guarantor under this Guaranty, whether now known or hereafter learned by Buyer, it being understood that Guarantor is capable of and assumes the responsibility for being and remaining informed as to all such facts and circumstances;

(g) until all Obligations in Default have been fully paid and/or performed, any rights of subrogation, contribution, reimbursement, indemnification, or other rights of payment or recovery for any payment or performance by it hereunder. For the avoidance of doubt, if any amount is paid to Guarantor in violation of this provision, such amount shall be held by Guarantor for the benefit of, and promptly paid to, Buyer; and

(h) all defenses based on suretyship, including any defense arising from a matter waived under this Guaranty.

Notwithstanding the foregoing or anything else herein, Guarantor does not waive and shall be entitled to assert any defenses, claims or other rights that are available to Seller (other than bankruptcy or insolvency) and all limitations of liability under the PPA shall also apply to this Guaranty.

10. <u>Cumulative Remedies</u>. The rights and remedies of Buyer hereunder shall be cumulative and not alternative to any other rights, powers, and remedies that Buyer may have at law, in equity, or under the PPA. The obligations of Guarantor hereunder are independent of those of Seller and shall survive unaffected by the bankruptcy of Seller. Buyer need not join Seller in any action against Guarantor to preserve its rights set forth herein.

11. <u>Representations and Warranties</u>. Guarantor represents and warrants to Buyer as follows:

(a) Guarantor is a [corporation,] duly organized, validly existing, and in good standing under the laws of the state of its incorporation. [Seller is a direct or indirect wholly-owned subsidiary of Guarantor.] Guarantor has all necessary corporate power and authority to execute and deliver this Guaranty and to perform its obligations hereunder.

(b) The execution, delivery and performance of this Guaranty has been duly and validly authorized by all corporate proceedings of Guarantor and is not in violation of any law, judgment of court or government agency. This Guaranty has been duly and validly executed and delivered by Guarantor and constitutes a legal, valid and binding obligation of Guarantor, enforceable against Guarantor in accordance with its terms.

12. <u>Collection Costs</u>. Guarantor hereby agrees to pay to Buyer, upon demand, and in addition to the maximum liability set forth in Section 2 hereof, all reasonable attorneys' fees and other expenses which Buyer may expend or incur in enforcing the Obligations against Seller and/or enforcing this Guaranty against Guarantor, whether or not suit is filed, including, without limitation, all attorneys' fees, and other expenses incurred by Buyer in connection with any insolvency, bankruptcy, reorganization, arrangement, or other similar proceedings involving Seller that in any way affect the exercise by Buyer of its rights and remedies hereunder.

13. <u>Severability</u>. Should any one or more provisions of this Guaranty be determined to be illegal or unenforceable, all other provisions nevertheless shall be effective.

14. <u>Waiver or Amendment</u>. No provision of this Guaranty or right of Buyer hereunder can be waived, nor can Guarantor be released from Guarantor's duties hereunder, except by a writing duly executed by Buyer. This Guaranty may not be modified, amended, revised, revoked, terminated, changed, or varied in any way whatsoever except by the express terms of a writing duly executed by Buyer.

15. <u>Successors and Assigns</u>. Guarantor shall not assign its rights or obligations under this Guaranty without the prior written consent of Buyer, and any assignment without such prior written consent shall be null and void and of no force or effect. This Guaranty shall inure to the benefit of and bind the successors and permitted assigns of Buyer and Guarantor.

16. <u>Entire Agreement</u>. This Guaranty reflects the whole and entire agreement of the parties and supersedes all prior agreements related to the subject matter hereof.

17. <u>Governing Law</u>. The interpretation and performance of this Guaranty and each of its provisions shall be governed and construed in accordance with the laws of the State of California. The parties hereby submit to the exclusive jurisdiction of the courts of the State of California, and venue is hereby stipulated as San Francisco, California.

18. <u>Termination</u>. Guarantor may terminate this Guaranty by providing written notice of such termination to Buyer; <u>provided</u>, <u>however</u>, that such termination by Guarantor shall not be effective unless and until Guarantor has provided replacement Seller Credit Support (as defined in the PPA) in accordance with Article 8 of the PPA. No such termination shall affect Guarantor's liability with respect to any Obligations incurred under the PPA prior to the time the termination is effective, which liabilities remain guaranteed pursuant to the terms of this Guaranty.

19. <u>Notices</u>. All notices, requests, claims, demands, and other communications hereunder shall be in writing and shall be given (and shall be deemed to have been duly given upon receipt) by delivery in the manner contemplated by the PPA, addressed as follows:

(a) <u>if to Buyer</u> as provided in the PPA

(b) <u>If to Guarantor</u>

Attn:		
Phone:		

With a copy to:

Attn:		
Phone:		

or to such other address(es) as the person to whom notice is given may have previously furnished to the others in writing in the manner set forth above.

IN WITNESS WHEREOF, the Guarantor has caused this Guaranty to be duly executed and delivered by its duly authorized representative as of the day and year first above written.

[Guarantor]

By: _____ Name: Title:

EXHIBIT 1

QUARTERLY/MONTHLY REPORTS

Each quarterly/monthly report shall include the following items:

- 1. Cover page.
- 2. Brief description of the Facility.
- 3. Site plan of the Facility.
- 4. Description of any material planned changes to the Facility.
- 5. Table showing critical path schedule of major items and activities.
- 6. Summary of major activities in the preceding quarter/month.
- 7. Forecast of major activities scheduled for the current quarter/month.
- 8. Forecast of Commercial Operation Date.

EXHIBIT 2

REQUIRED INSURANCE

Type of Insurance	Minimum Limits of Coverage
Commercial General Liability and Excess or Umbrella policies combined limits.	\$11,000,000 each occurrence and \$11,000,000 general aggregate

Commercial General Liability ("CGL") insurance shall cover liability arising from premises, operations, products/completed operations, bodily injury and property damage, personal injury and advertising injury, sudden and accidental pollution, and liability assumed under an insured contract (including the tort liability of another assumed in a business contract), all with limits as specified above. There shall be no endorsement or modification of the CGL insurance limiting the scope of coverage for liability arising from explosion, collapse, or underground property damage. Sudden and accidental pollution coverage may be placed separately with limits equal to the combined policy limited noted above.

Buyer shall be included as an insured under the CGL policy, using ISO additional insured endorsement CG 20 10 (or a substitute providing equivalent coverage), and under the commercial umbrella or excess insurance. The commercial umbrella or excess insurance shall provide coverage over the top of the CGL insurance, the Business Automobile Liability insurance, and the Employer's Liability insurance.

The CGL and commercial umbrella or excess insurance to be obtained by or on behalf of Seller shall be endorsed or shall otherwise provide as follows:

The CGL insurance as afforded by this policy for the benefit of Buyer shall be primary as respects any claims, losses, damages, expenses, or liabilities arising out of this Agreement, and insured hereunder, and any insurance carried by Buyer shall be excess of and noncontributing with insurance afforded by this policy. Commercial umbrella or excess insurance shall be non-contributory.

Seller shall not be required to have the above mentioned liability insurance coverages until the Commercial Operation Date.

Business Automobile Liability	\$1,000,000 each accident, including all Owned (if any), Non-Owned, Hired and Leased Autos Business Automobile Liability insurance shall be written on ISO form CA 00 01, CA 00 05, CA 00 12, CA 00 20, or a substitute form providing equivalent liability coverage.
Workers Compensation	Statutory Requirements. Seller may comply with these requirements through the use of a qualified self-insurance plan.
Employers Liability	\$1,000,000 bodily injury by accident - each accident; \$1,000,000 bodily injury by disease - each employee; \$1,000,000 bodily injury by disease-policy limit.

Builder's Risk	No later than the date construction begins, Builder's Risk insurance, or an installation floater, shall be written on an "all risk" basis and include earthquake, flood, collapse, hot testing and debris removal. Coverage shall be written on a replacement cost basis for physical damage with reasonable sublimits. The Policy Limits will be based upon Maximum Foreseeable Loss (MFL) which is supported by an engineering assessment or an amount no less than 75% of the hard costs of the Facility.
All-Risk Property insurance	Beginning on the Commercial Operations Date, All- Risk Property insurance on a replacement costs bases including coverage for: (i) fire, flood, wind storm, tornado and earthquake and (ii) Machinery Breakdown insurance; all subject to reasonable sublimits. The Policy Limits will be based upon Maximum Foreseeable Loss which is supported by an engineering assessment or an amount no less than 75% of the hard costs of the Facility.
Business Interruption insurance	Amount required to cover Seller's continuing expenses resulting from full interruption for a period of twelve (12) calendar months.
	Business Interruption insurance shall cover loss of revenues attributable to the Facility by reason of total or partial suspension or delay of, or interruption in, the operation of the Facility as a result of an insured peril covered under Property insurance as set forth above, to the extent available on commercially reasonable terms as determined by Buyer, subject to a reasonable deductible which shall be the responsibility of Seller. Notwithstanding any other provision of this Agreement, Seller shall not be required to have Business Interruption insurance until the Commercial Operation Date.

EXHIBIT B

FORM OF AUTHORIZING RESOLUTION OF SOLAR PARTICIPANT

RESOLUTION OF THE [GOVERNING BODY] OF THE [SOLAR PARTICIPANT] APPROVING AND AUTHORIZING THE EXECUTION OF A POWER PURCHASE CONTRACT BETWEEN THE SOLAR PARTICIPANT AND THE MUNICIPAL ELECTRIC AUTHORITY OF GEORGIA, THE PLEDGE OF THE FULL FAITH AND CREDIT OF THE SOLAR PARTICIPANT TO SECURE ITS PAYMENT OBLIGATIONS THEREUNDER, AND FOR SUCH OTHER PURPOSES.

WHEREAS, pursuant to the Municipal Electric Authority Act (the "Act"), the [Solar Participant] (the "Solar Participant") has previously entered into one or more Power Sales Contracts (each, as amended, a "Power Sales Contract") with the Municipal Electric Authority of Georgia (the "Authority") for provision of the Solar Participant's bulk electric power supply needs by the Authority from defined projection projects and sources; and

WHEREAS, under one such Power Sales Contract, the Project One Power Sales Contract (the "**Project One Power Sales Contract**"), the Authority further agreed to provide or cause to be provided additional power needs of the Solar Participant in excess of its entitlement to power supplied under the Project One Power Sales Contract ("**Supplemental Power**"); and

WHEREAS, the Project One Power Sales Contract provides that the Solar Participant may elect to procure an alternate source of Supplemental Power other than that provided by the Authority from the output of an Authority project; and

WHEREAS, the Authority adopted a Supplemental Power Policy (the "**Supplemental Power Policy**") under which the Solar Participant and the Authority may make elections regarding provision and procurement of Supplemental Power; and

WHEREAS, the Solar Participant has determined that, in order to meet the growing and diverse energy needs of its customers, it has need for an additional type of economical, reliable source of electric power and energy beyond that provided from the sources available resources of the Authority under the Project One Power Sales Contract and other contracts between the City and the Authority; and

WHEREAS, the Authority has informed the Solar Participant that the Authority has an opportunity to procure a substantial amount of Supplemental Power for a multi-year term through a Power Purchase Agreement with Pineview Solar LLC (the "Company") for the output and services of approximately 80 MWac from a photovoltaic solar energy generation facility located in Wilcox County, Georgia (the "Facility") to be constructed, owned, operated, and maintained by the Company (such agreement, the "Supplemental Power Purchase Agreement" or "SPPA"); and

WHEREAS, in accordance with the Supplemental Power Policy, the Solar Participant has requested that the Authority purchase from the Company power, output and services of the Facility to cause to be provided to the City its Supplemental Power; and

WHEREAS, the Authority has agreed to cause to be provided the Solar Participant's Supplemental Power from the power, output and services of the Facility pursuant to the terms of a Power Purchase Contract (the "**PPC**") in substantially the form attached as <u>Exhibit A</u> hereto; and

WHEREAS, the Solar Participant finds, and the Solar Participant and the Authority agree that the PPC is supplemental to, and is authorized by, the Project One Power Sales Contract and that the Products (as defined in the SPPA) constitute Supplemental Power as defined in the Supplemental Power Policy; and

WHEREAS, the Solar Participant determines that the Solar Participant's payment obligations for Supplemental Power under the PPC authorized thereby shall constitute the general obligations of the Solar Participant for the payment of which the full faith and credit of the Solar Participant is pledged, obligating the Solar Participant to provide for the assessment and collection of an annual tax sufficient in amount to provide funds annually to make all payments due thereunder; and

WHEREAS, the [Governing Body] desires to approve the PPC; to authorize the execution and delivery of the PPC and other such documents, certificates, and opinions described therein; and authorize such further actions as necessary for the Solar Participant to procure Supplemental Power as provided thereby.

NOW, THEREFORE, BE IT RESOLVED by the [Governing Body] of the Solar Participant as follows:

1. <u>Incorporation of Recitals</u>. The recitals set forth above are hereby incorporated in the body of this Resolution.

2. <u>Findings and Determinations</u>. All findings and determinations contained in the PPC, including the recitals thereto, are hereby incorporated herein by reference, and are hereby adopted as findings and determinations of the [Governing Body] of the Solar Participant.

3. <u>Defined Terms</u>. All capitalized terms used herein but not otherwise defined herein shall have the meanings set forth in the PPC.

4. <u>Supplemental Power Purchase Agreement</u>. The [Governing Body] of the Solar Participant acknowledges receipt of the form of the SPPA to be executed by the Authority and Company.

5. <u>Authorization to Execute PPC</u>. The [Governing Body] of the Solar Participant hereby authorizes the Solar Participant to enter, as a Solar Participant (defined therein) into the PPC in substantially the form attached as <u>Exhibit A</u> hereto, and to perform the same, and the [Title of Officer] of the Solar Participant is hereby authorized on behalf of the Solar Participant to execute and deliver the PPC. The [Title of Officer], with the advice of Counsel to the Solar Participant, is authorized to agree to such changes to the PPC as may be necessary prior to execution thereof, and the execution and delivery of the PPC shall be conclusive evidence of such approval. The [Title of Officer] of the Solar Participant is authorized to attest the execution by the [Title of Officer] of the PPC and to affix the seal of the Solar Participant to such documents.

6. <u>Further Authority</u>. The [Governing Body] hereby authorizes, empowers and directs the _____] and any necessary representatives of the Solar Participant to do all such acts and things and to execute all such documents as may be necessary to carry out and comply with the provisions and intent of this Resolution and the PPC.

7. <u>Authorized Representative</u>. The [Title of Officer] and [Title of Officer] of the Solar Participant are each hereby each designated as Authorized Representatives of the Solar Participant, and may execute notices, certificates, requests, estimates and other documents contemplated by the PPC, subject to the limitations contained herein.

8. <u>Repeal of Conflicting Resolutions</u>. All resolutions and parts of resolutions in conflict with this Resolution are hereby repealed to the extent of such conflict.

9. <u>Effective Date</u>. This Resolution (including the recitals first above written, which are hereby incorporated into this Resolution) shall take effect immediately upon its adoption; a copy of this Resolution may be filed in such offices as the undersigned or such development authority may elect to file this Resolution. All resolutions, or parts of resolutions, in conflict herewith are repealed.

[SIGNATURES BEGIN ON FOLLOWING PAGE]

DULY ADOPTED at a meeting of the [Governing Body] of the Solar Participant, held this _____ day of _____ 2021.

Solar Participant

By:		
Name:		
Title:		

ATTEST:

By:	
Name:_	
Title:	

[SEAL]

EXHIBIT A

FORM OF PPC

[FORM ATTACHED]

CERTIFICATE OF CLERK

The undersigned, being the duly appointed, qualified, and acting Clerk of the [Governing Body] of the Solar Participant, **DOES HEREBY CERTIFY** that the foregoing pages of typewritten matter constitute a true and correct copy of a resolution adopted on _______, 2021, by the [Governing Body] of the Solar Participant in a meeting duly called and assembled, after due and reasonable public notice was given in accordance with the procedures of the Solar Participant and with the applicable provisions of law, which was open to the public and at which a quorum was present and acting throughout, and that the original of such resolution appears of public record in the minute books of the [Governing Body] of the Solar Participant, which are in my custody and control.

I do hereby further certify that all members of the [Governing Body] were present at said meeting except the following members who were absent:

and that the resolution was duly adopted by the following vote:

The following voted "Aye":	
The following voted "Nay":	· · · · · · · · · · · · · · · · · · ·
The following Did Not Vote:	,
	·

WITNESS my hand and the official seal of the [Solar Participant], this _____ day of _____, 2021.

Clerk

[SEAL]

EXHIBIT C

FORM OF OPINION OF COUNSEL TO SOLAR PARTICIPANT

[TO BE REPRINTED ON LETTERHEAD OF SOLAR PARTICIPANT'S COUNSEL]

, 2021

Municipal Electric Authority of Georgia Atlanta, Georgia

[Solar Participant] [Address], Georgia

Seyfarth Shaw LLP Atlanta, Georgia

RE: Power Purchase Contract between the Municipal Electric Authority of Georgia and [Solar Participant]

Ladies and Gentlemen:

[I/We] have acted as Counsel to [Solar Participant] (the "**Solar Participant**") preliminary to and in connection with the authorization and execution of the above-captioned Power Purchase Contract, dated as of ______, 2021 (the "**PPC**"), between the Municipal Electric Authority of Georgia (the "**Authority**") and the Solar Participant. In so acting, we have examined such documents and matters of law as we have considered desirable to render the opinions hereinafter expressed, including but not limited to the following:

(a) The Constitution of the State of Georgia of 1983, particularly Article IX, Section III, Paragraph I(a) thereof and various acts of the General Assembly of Georgia relating to the Solar Participant;

- (b) The PPC;
- (c) The Project One Power Sales Contract;

(d) The validation certificate provided pursuant to the order issued by the Superior Court of Fulton County in *State of Georgia v. Municipal Elec. Auth. of Georgia et al.* Civil Action File No. 2018CV37032 (Fulton Co. Sup. Ct. July 17, 2018);

(e) Minutes of the meeting of the [Governing Body] of the Solar Participant held on ______, 2021, authorizing at such meeting the execution of the PPC by the Solar Participant; and

(f) Such other documentation and matters of law as [I/we] have deemed necessary.

Whenever [I/we] have stated that [I/we] have assumed any matter, it is intended to indicate that we have assumed such matter without making any factual, legal or other inquiry or investigation, and without expressing any opinion or conclusion of any kind, concerning such matter. [I/we] assume no issue of unconstitutionality or invalidity of a relevant law unless a reported case has so held.

Reference is made to the opinion dated the date hereof of Seyfarth Shaw LLP, Atlanta, Georgia, as counsel to the Authority, upon which [I/we] have relied, with your permission, with respect to all matters related to the validity and enforceability of the Supplemental Power Purchase Agreement (the "SPPA") between the Authority and Pineview Solar LLC and the security pledged thereunder. [I/we] have reviewed sufficient information to assume that the Project One Power Sales Contract between the Solar Participant and the Authority has been judicially confirmed and validated by order of the Superior Court of Fulton County, Georgia. [I/we] have further assumed, in reliance upon the opinion of Seyfarth Shaw LLP, that the Authority has all requisite power and authority to enter into and perform its obligations under the SPPA, and that the SPPA is a valid and binding agreement, enforceable against the Authority in accordance with its terms.

Based upon the foregoing, it is [my/our] opinion that:

1. The PPC has been duly and validly authorized, executed and delivered by the Solar Participant and the provisions thereof which obligate the Solar Participant are legal, valid and binding obligations of the Solar Participant enforceable in accordance with the terms thereof. Under the terms of the PPC, the Solar Participant is obligated to levy a tax, at a rate sufficient, as described in the PPC, on all property in the Solar Participant's jurisdiction subject to such tax, to the extent necessary to generate sufficient revenue to pay its obligations under the PPC.

2. To the best of [my/our] knowledge and belief after reasonable inquiry, the PPC and the performance of the Solar Participant's obligations thereunder will not conflict with or be in violation of any applicable federal, state, or local law or ordinance or, to the best of [my/our] knowledge and belief, be in violation of, or constitute a default under, any agreement or instrument to which the Solar Participant is party or by which the Solar Participant is bound.

3. Each [officer/official] of the Solar Participant who executed the PPC was on the date of the execution thereof, and is on the date hereof, the duly, elected or appointed qualified incumbent of his or her office.

4. The notices given prior to each meeting of the Solar Participant at which any action was taken relating to the PPC and the security therefor comply with the applicable notice requirements of Georgia law, and said meetings were conducted in accordance with all other applicable requirements of Georgia law.

5. There is no action, suit, proceeding, inquiry or investigation, at law or equity, by or before any court or public board or body pending or, to the best of [my/our] knowledge and belief, after making due inquiry with respect thereto, threatened against or affecting the Solar Participant,

nor to [my/our] knowledge is there any basis therefore, which in any way questions the creation or existence of the Solar Participant or the powers of the Solar Participant, or which might result in a material adverse change in the condition (financial or otherwise), business or affairs of the Solar Participant or wherein an unfavorable decision, ruling or finding would adversely affect the validity or enforceability of the PPC or any other agreement or instrument to which the Solar Participant is a party and which is used or contemplated for use in connection with the consummation of the transactions contemplated by the PPC or which in any way would adversely affect the levy or collection of taxes by the Solar Participant to fulfill its obligations pursuant to the PPC.

6. All consents, approvals or authorizations, if any, of any governmental authority or agency or other person required on the part of the Solar Participant in connection with the approval of the PPC, the execution and delivery of the same and the consummation of the transactions contemplated thereby have been obtained, and the Solar Participant has complied with any applicable provisions of law requiring any designation, declaration, filing, registration and/or qualification of the Solar Participant with any governmental authority or agency or other person in connection with such execution, delivery and consummation.

The foregoing opinions are qualified to the extent that the enforceability of the PPC might be limited by (i) bankruptcy, fraudulent transfer, moratorium, insolvency, reorganization, or other laws affecting the enforcement of creditors' rights, (ii) limitations imposed by general principles of equity upon specific enforcement, injunctive relief or other equitable remedies, (iii) the exercise of judicial discretion in appropriate cases and (iv) to the following qualifications:

[I/We] express no opinion as to the validity or enforceability of any of the (a) following provisions that may be contained therein: (i) any provisions which purport to waive any defense, counterclaim, set off or deduction arising from any violation of applicable federal or state securities or usury laws, any fraud on the part of any other party, any failure to give notice of a disposition of collateral to the extent required under applicable law, any disposition of collateral other than in a commercially reasonable manner, or the effect of any applicable statute of limitation, (ii) any choice of law provisions therein, (iii) any provisions which purport to waive the right to trial by jury or purport to consent to or waive any objection to the jurisdiction or venue of any particular court, and (iv) any provisions which provide for payment of interest on unpaid interest or which, due to prepayment, acceleration, or otherwise, would cause the rate of interest to exceed five percent (5.0%) per month. [I/We] also note that any provisions requiring any party to pay the attorneys' fees of any other party may be subject to compliance with applicable legal requirements and limitations and that the provisions thereof may be subject to the effect of the provisions of law regarding mutual departures from strict contractual terms. Nothing in this paragraph (a) is intended to limit any of the other qualifications or exceptions to [my/our] opinions set forth in this letter.

(b) Enforcement of any warranties and indemnities contained therein may be limited by applicable federal or state securities laws as violations of public policy and may be limited to the extent such indemnities would require any party to indemnify another party for costs, losses, liabilities, claims, damages or expenses incurred by or asserted against such party as a result of action or inaction of such party constituting negligence. In addition, it is possible that a court would not enforce any warranties or indemnities with respect to environmental matters contained therein.

(c) With respect to the enforceability thereof, [I/we] have assumed that, to the extent that applicable law would require the rights and remedies set forth therein to be exercised in good faith or in a reasonable or commercially reasonable manner as a condition to the enforceability thereof, the persons having remedial rights thereunder will observe and satisfy such legal requirements.

The undersigned's engagement as Counsel to the Solar Participant imposed no duty upon the undersigned to undertake any due diligence investigation as to either: (i) the adequacy of the security for the PPC, (ii) the business or financial condition of the Solar Participant, or (iii) the veracity of any representations or certification made by the [Solar Participant] on which [I/we] have relied. No opinion is expressed as to the federal or state tax-exempt status of the obligations or the interest thereon, or the applicability of the federal securities laws or the Blue Sky laws of any state with respect to the PPC.

The opinions set forth herein are limited to the laws of the State of Georgia and applicable federal laws. The opinions represent [my/our] legal judgment based upon our review of the law and the facts that we deem relevant to render such opinion and is not a guarantee of a result.

This opinion letter is rendered as of its date, and [I/we] express no opinion as to circumstances or events that may occur subsequent to such date. Further [I/we] undertake no, and hereby disclaim any, obligation to advise you, or any other person permitted to rely hereon, of any change in applicable law or relevant facts or any new development which might affect any matters or opinions set forth herein.

This opinion letter is given solely for the benefit of the addressees and their successors and assigns. This opinion letter is not intended to be employed in any transaction other than the one described above and is being delivered to the addressees with the understanding that it may not be published, quoted, relied on or referred to by, and copies may not be delivered or made available to, in whole in or part, any other person or entity (other than the addressees' counsel or any applicable rating agency) or used for any other purpose with the express prior written consent of this firm in each instance.

Very truly yours,